Kamila Michałowska

## Sensitivity analysis of parameters influencing the carbon footprint of Brazilian office buildings

Master's thesis in Industrial Ecology Supervisor: Edgar Hertwich June 2020

Master's thesis

NTNU Norwegian University of Science and Technology Faculty of Engineering Department of Energy and Process Engineering



Kamila Michałowska

## Sensitivity analysis of parameters influencing the carbon footprint of Brazilian office buildings

Master's thesis in Industrial Ecology Supervisor: Edgar Hertwich June 2020

Norwegian University of Science and Technology Faculty of Engineering Department of Energy and Process Engineering



## Abstract

Energy and material use associated with buildings causes significant environmental impacts which need to be urgently reduced, especially in emerging economies such as Brazil, whose building stock is expected to significantly grow in the following years. Office buildings are typically characterized by higher energy use than residential buildings, yet they have a relatively poor coverage in the research. So far, there has been no comprehensive assessment of the possible range of climate change impacts caused by Brazilian office buildings.

Therefore, the purpose of this study was to investigate the carbon footprint of Brazilian office buildings. Building archetype definition served as a basis for the model, which included 10 variable parameters. To fully explore the possible combinations of these parameters' values, Latin hypercube sampling was performed. The selected buildings were simulated in building energy software in order to estimate their energy performance. The building model that was used during energy simulations was integrated with material composition data to properly reflect energy-material interactions. Further, a life cycle assessment (LCA) study enabled to estimate GWP100 impacts of the sampled buildings. The analysis included the building life cycle stages associated with material production and transport (A1-A4), construction (A5), replacement (B4), operational energy use (B6) and end-of-life stage (C).

The results of the model have shown variations in the total GWP values from 20 to 108 kg  $CO_2$ -eq/m<sup>2</sup>/year, with the biggest contribution from operational energy use phase. The GWP impacts are also influenced by emissions associated with initial and recurrent material demand, both dominated by aluminum. There was also a significant contribution of paint to the replacement emissions.

The most important parameters for GWP impacts were found to be electricity mix, climate, cooling efficiency and window effective opening area (for mixed-mode buildings). Based on these results, recommendations for mitigation strategies are given.

# Acknowledgments

I would like to express my sincere gratitude to my supervisor, prof. Edgar Hertwich, and to Niko Heeren, for your support and insightful comments, which allowed me to better structure my work and to sharpen my research skills.

My appreciation also extends to all the people who helped me in my data collection efforts: Damasia Gonzalez dos Santos Oliveira, Pedrinho Goldman, Vanessa Gomes da Silva, Fernanda Belizário Silva, and Camila Kramel. I cannot underestimate your contribution to my understanding of the Brazilian construction industry. The data and comments you provided enabled me to gain confidence in my research choices.

Thanks also go to Andrea A. Nishtad, for our shared journey through the peculiarities of EnergyPlus and the construction industry. I also thank Ruslan Zhuravchak, for your useful tips on building energy research.

In addition, I would like to thank Karina Condeixa, Leticia Neves, Edward Borgstein, Roberto Lamberts, and Tatiana Alves. Your pioneering research on Brazilian buildings was the foundation for this thesis. Without you, this work would not have been possible.

And last but not least, I am grateful to Adam, for your unconditional support at all times.

# Table of Contents

	List of	Figu	res	х
	List of	Tabl	es	х
	List of	Abb	reviations	ĸi
1	Intro	oduc	tion1	2
	1.1	Con	text1	2
	1.2	Sco	pe restriction: office buildings in Brazil1	3
	1.3	Aim	of the study1	3
	1.4	The	sis overview1	4
2	Back	ιgroι	ınd1	5
	2.1	Braz	zilian context1	5
	2.2	Ene	rgy characteristics of Brazilian office buildings1	9
	2.3	Mate	erial characteristics of Brazilian office buildings2	2
	2.4	Emi	ssion characteristics of Brazilian office buildings2	3
	2.5	Sen	sitivity analysis in building research2	6
3	Meth	nodo	logy2	9
	3.1	Offic	ce building archetype definition3	0
	3.1.	1	Climate	0
	3.1.2	2	Geometry and construction materials3	0
	3.1.3	3	Windows and shading3	2
	3.1.4	4	HVAC system3	3
	3.1.	5	Internal heat gains	4
	3.1.0	5	Lifetime	4
	3.1.7	7	Archetype definition summary3	4
	3.2	Sam	ple choice	5
	3.3	Ene	rgy modeling3	6
	3.4	Mate	erial modeling3	7
	3.5	Emi	ssion modeling4	0
	3.5.3	1	Modules A1-A4: Production and transport to the building site4	1
3.5. 3.5. 3.5. 3.5.		2	Module A5: Installation into the building4	2
		3	Module B4: Replacement4	2
		4	Module B6: Operational energy use4	3
		5	Module C1: Deconstruction, demolition4	4
	3.5.0	5	Modules C2-C4: Waste processing and disposal4	4
	3.5.2	7	Emission modeling summary4	5
	3.6	Sen	sitivity analysis4	5

4	Res	ults	.48
	4.1	Energy modeling results	.48
	4.2	Material modeling results	.51
	4.3	Emission modeling results	.52
	4.4	Regression analysis	.58
5	Disc	cussion	.60
	5.1	Range of carbon footprint values	.60
	5.2	Importance of parameters influencing carbon footprint	.62
	5.3	Recommendations for better emission performance	.64
	5.4	Limitations of the study	.65
	5.5	Further research	.66
6	Con	clusions	.67
Re	feren	ces	.69
Ap	Appendices		

# List of Figures

Figure 1 Five regions of Brazil with the twelve most populous cities15
Figure 2 Schematic representation of the methodology
Figure 3 Material types according to the data processing procedure
Figure 4 Total energy load and its components: equipment, lighting, cooling, and heating.
Figure 5 Total energy load as a function of city49
Figure 6 Total energy load as a function of the chosen variable parameters50
Figure 7 GWP impacts of the building life cycle modules54
Figure 8 Relationship between the total GWP and the total energy load54
Figure 9 GWP of module A1-A4 as a function of material and building type55
Figure 10 GWP of module B4 as a function of material and building type55
Figure 11 GWP of module A1-A4 as a function of selected materials and building lifetime.
Figure 12 GWP of module B4 as a function of selected materials and building lifetime56
Figure 13 Total GWP impacts as a function of city56
Figure 14 Total GWP impacts as a function of the chosen variable parameters57

# List of Tables

Table 1 Three office building archetypes and their main characteristics	31
Table 2 Overview of the variable parameters for office building archetypes	35
Table 3 Life-cycle stages of construction products according to EN 15978	40
Table 4 Energy use for construction of office building archetypes	42
Table 5 Energy use for demolition of office building archetypes	44
Table 6 Independent variables used in the regression analysis	47
Table 7 Window-to-wall ratio and material intensity of glass and aluminum	51
Table 8 Results of the multivariate regression analysis of total energy loads	58
Table 9 Results of the multivariate regression analysis of GWP impacts	59

## List of Abbreviations

AC	Air conditioning
CDH	Cooling degree hours
CL-8	Office building archetype with 8 floors and a cellular layout
CL-16	Office building archetype with 16 floors and a cellular layout
СОР	Coefficient of performance
GHG	Greenhouse gas
GWP	Global warming potential
EUI	Energy use intensity
HVAC	Heating, ventilation, and air conditioning
LCA	Life cycle assessment
LCI	Life cycle inventory
LHS	Latin hypercube sampling
MM	Mixed-mode (ventilation or building)
OPL-8	Office building archetype with 8 floors and an open plan layout
NBR	Brazilian Regulatory Standard
SGHC	Solar heat gain coefficient
WWR	Window-to-wall ratio

## 1 Introduction

### 1.1 Context

Buildings are an indispensable part of human society, ensuring protection from the outside world and allowing for comfort despite varying weather conditions. At the same time, satifying these needs is associated with flows of materials and energy throughout life cycle stages of a building: construction, operation and demolition. In this system of close interdependencies, flows of material at the construction stage determine the magnitude of future material flows needed to maintain the building, and energy flows needed to provide services to its inhabitants during the operational phase. At the end of the building lifetime, materials are dismantled and disposed of as waste or sent off for recycling. During all of these life cycle stages, greenhouse gas (GHG) emissions are emitted, contributing to climate change impacts.

Buildings are responsible for a significant part of global GHG emissions, both through direct emissions (occurring directly in this sector) and indirect emissions occurring in other sectors due to demand arising from building-related activity. Considering the energy sector, as much as half of global electricity consumption could be attributed to buildings. (IPCC, 2015) Demand from the industry sector includes construction materials such as cement and steel, which are particularly crucial as these materials are emission-intensive. These emissions are inherent to the chemical production processes so they cannot be addressed merely by measures such as energy efficiency or decarbonization of the energy sector. (Material Economics, 2018) Buildings are therefore one of the biggest contributors to anthropogenic climate change but, at the same time, can be seen as the key to mitigation strategies.

Developing countries should be the focus of mitigation strategies in the construction sector, as their floor area is expected to more than double within the next 40 years. (IEA, 2017) The ongoing accumulation of building and infrastructure stock in these countries is associated with significant material emissions. (Müller et al., 2013) Long lifetime of such structures means that potentially inefficient designs create carbon-intensive emission pathways that last for years, which is known as the lock-in effect. (Seto et al., 2016)

Many energy efficiency opportunities in buildings are cost-saving, and additionally, emerging economies could also profit through non-monetary gains such as energy security, reduced pollution, and improved productivity. Delaying the action by another 10 years would lead to significant amounts of additional CO<sub>2</sub> emissions from unnecessary energy demand, associated also with significant additional financial costs. (IEA, 2019a)

Reduction of material use in buildings, besides emission savings, offers co-benefits related to decreased mining activity which often destroys natural ecosystems and impacts livelihoods of local people, such as in case of mining-induced deforestation (Sonter et al., 2017) or illegal sand mining in rivers (Bendixen et al., 2019).

As can be seen, there is an urgent need to develop strategies for creating efficient building stock. These strategies should identify the most beneficial opportunities from the life cycle perspective, which requires in-depth understanding of the existing stock and construction

practices. In many developed countries, this need had been identified and have given rise to projects such as TABULA – a European project intended to characterize the residential stock of chosen countries by means of differentiation into building types of various energy standards and energy needs. (Institut Wohnen und Umwelt GmbH, 2017) In developing countries, data availability issues create a challenge for the building stock research, making it difficult to assess possible emission savings and to implement the most efficient designs. This shows the importance of further efforts aimed at improving the understanding of the building stock in emerging economies. However, the stock characterization is faciliated if the scope is restricted to a given building type and a given country.

## 1.2 Scope restriction: office buildings in Brazil

Building types can be divided into two main categories: residential and non-residential. Non-residential construction is more difficult to address as it is less homogenous, with buildings providing various end-uses and thus possessing different characteristics of material and energy use. Non-residential buildings are often divided into commercial and public services sectors. Offices are the most important group of commercial buildings in developed countries, for example in the U.S. they account for around half of the total number of commercial buildings and half of the total commercial floorspace. (U.S. EIA, 2015) They are also one of the two fastest growing commercial building stock types, the other one being warehouses. (U.S. EIA, 2015) If such trends can be observed in developed countries such as the U.S., the current and expected future increase in office floorspace is likely even more pronounced in emerging economies, where economic growth is often associated with increasing importance of the services sector. Previous studies have shown that office floorspace per capita generally increases with the growth in services value added per capita. (Deetman et al., 2020) Office building stock can be of high energy intensity, especially in places where warm climate in conjuction with internal heat load characteristics of offices create the need for mechanical cooling systems. As it has been pointed out by other authors, there is inferior coverage of non-residential construction in building research. (Deetman et al., 2020; Maslesa et al., 2018) Office buildings, being such a distinct group of non-residential buildings, are therefore an important research area to focus on.

The spatial scope can be restricted to Brazil. It is the third most populous country among major emerging national economies, known as BRICS (Brazil, Russia, India, China, and South Africa). China and India, the two most populous of the BRICS countries, have received relatively more attention in building stock research compared to Brazil, which can be verified by a few simple web search queries using a search engine such as Google Scholar. Brazilian commercial floor space is expected to increase by about 60% between 2000 and 2050. (ICCA, 2012) Together with other developing countries, it forms the key drivers of increasing global demand for materials, such as steel and cement. (IEA, 2019b) Therefore, a valuable contribution can be made by addressing any existent gaps in the current state of Brazilian building stock research.

### 1.3 Aim of the study

There has been a number of studies concerning office buildings in Brazil, but research investigating their environmental impact is scarce. In particular, the importance of factors influencing the greenhouse gas emissions is not fully understood.

The purpose of this work is therefore to provide an empirical characterization of office buildings, their composition, construction, and energy use. Further, the modeling work

should assess the implementation of improvement options, including through improved design. The assessment should take into account the life cycle GHG emissions. A good documentation of empirical observations, modeling assumptions, modeling methods, code and data are required.

This work intends to answer the following questions:

- 1. What is the possible range of the carbon footprint values for Brazilian office buildings?
- 2. Which parameters have the most impact on the carbon footprint?
- 3. What can be done to improve the emission performance of the current office building stock in Brazil?

The study will focus on selected parameters and their influence on the carbon footprint, understood as the total GHG emissions caused by the building during its whole life cycle, from cradle to grave. The selected parameters will be chosen based on available literature. The parameter space will be explored using appropriate sampling methods and the carbon footprint values will be calculated based on life cycle assessment methodology, using GWP100 as metrics.

This research contributes to the current state of knowledge by quantifying the likely range of GHG emissions of Brazilian office buildings and identifying features that have the biggest influence on the emissions from the life cycle perspective. It intends to lay the groundwork for possible future studies on Brazilian office building stock, allowing for scenario modeling, and assisting in deployment of low-emission pathways.

#### 1.4 Thesis overview

The thesis is organized into 6 chapters, including this one, which presented the problem and its context, stated research questions and defined the scope of work.

The following, Chapter 2, provides more detail to the context of the study by describing the current issues related to Brazilian office building stock and the current state of knowledge. The available literature is used to specify which parameters are likely to have a significant contribution to carbon footprint of the buildings.

Chapter 3 explains the methods used to define the office building archetypes, calculate their energy performance and quantify GHG emissions. The chapter includes a description of the sample selection and sensitivity analysis methods.

Chapter 4 presents results of the model, showing the importance of variable parameters on carbon footprint values.

Chapter 5 includes a discussion of the results, particularly comparing them to existent literature on the subject of energy and emission performance research. Limitations of the study are acknowledged at this point.

Chapter 6 concludes the whole work.

# 2 Background

This chapter intends to put the research goals in a wider context. At first, the background necessary to understand the Brazilian reality is explained: demography, climate, energy sector, construction sector, and environmental issues. Further, Brazilian office buildings are characterized in terms of energy, materials, and emissions, based on available literature. Finally, research on sensitivity analysis in buildings is introduced in order to show the current state of knowledge on parameters influencing energy and emission performance of office buildings.

#### 2.1 Brazilian context

Brazil's population was estimated at around 211 million in 2019, with projections of 229 million in 2050. (The World Bank, 2019) Figure 1 shows the five main regions of Brazil and the twelve most populous cities according to the Brazilian Institute of Geography (IBGE). (IBGE, 2011) The map also shows the division to 27 states – each of the 12 cities is the biggest city in its own state. (IBGE, 2011)

Most of the population resides in the south and south-east of the country. According to IBGE, among all the Brazilian states, just 4 neighboring ones in the south and south-east of the country concentrate half of the working population (São Paulo, Minas Gerais, Rio de Janeiro, and Paraná). (IBGE, 2019a)



Figure 1 Five regions of Brazil with the twelve most populous cities. Adapted from work by Felipe Menegaz, used under CC-BY-SA 3.0.

According to the Köppen classification, the Brazilian climate types range significantly: a very humid rainforest climate in the Amazon region in the northwest, savanna in the center-west, dry semi-arid climate in the northeast, humid subtropical climate in the south. The most populated regions are the ones with humid subtropical climate in the south-east. These climatic conditions create a predominant need for cooling, while heating is mainly required in the southernmost part of the country for a very limited share of the year.

The electricity production in Brazil is based on hydropower, which accounted for 65% of electricity production in 2018. (MME, 2019) The fact that the national electricity production is based on renewable energy has made some people disregard the importance of energy efficiency measures. However, as reported by the Brazilian Ministry of Mines and Energy (MME) (2019), the importance of thermal power plants has been rising in the past decades, with natural gas being their main primary energy source. The emission intensity of Brazilian electricity mix is growing, and such a trend can only be altered by renewable energy development together with efforts to limit the growth of national electricity consumption. Particularly the further development of wind energy can be of help – although its share is still small, it has been steadily increasing in the recent years. (MME, 2019)

Until now, the electricity demand in Brazil has been rising, and some sectors are becoming more energy demanding than others. That is the case for the commercial building sector, whose share of national electricity consumption has grown in the past years. (MME, 2019) There is no official data on the structure of Brazilian commercial building stock, but office buildings are likely the biggest group of commercial buildings, as a similar relation is observed in other countries, such as the U.S. (U.S. EIA, 2015) The increasing energy use in commercial buildings, particularly office buildings, can be attributed to increasing office floor space and increasing need for cooling. These trends are also likely to continue in the future.

Commercial floor space in Brazil is predicted to increase by about 60% between 2000 and 2050. (ICCA, 2012) Increasing office floor space may also be reflected by the apparent recent increasing trend in the amount of people employed in office-related professions such as IT, communication, finance, real estate, professional and administrative activities. (IBGE, 2019a) Interestingly, almost half of the people employed in these professions were based either in Rio de Janeiro or São Paulo state. (IBGE, 2019a) Such statistics suggest that most of the national office floor area is concentrated in just a few cities situated in the south-east of the country, mainly São Paulo and Rio de Janeiro. This can be confirmed by studies such as the energy benchmarking study of corporate office buildings performed by Lamberts et al. (2015), as buildings situated in São Paulo and Rio de Janeiro were 75% of the total sample of 249 buildings; the rest were distributed among other cities, mostly Brasília, Florianópolis, Curitiba, Salvador and Porto Alegre.

As for the need of cooling, there are no national statistics on electricity end-uses of office buildings, but many studies show that cooling is often the single largest electricity end-use in Brazilian office buildings. (Alves et al., 2017; Borgstein and Lamberts, 2014; Carvalho et al., 2010; Lamberts et al., 2015; Pasquali et al., 2011) Some authors note that modern office buildings are often designed with a focus on aesthetics and with little concern for the building's energy performance, giving rise to buildings with high energy consumption. (Lima, 2010; Neves and Marques, 2017; Tamanini Junior and Ghisi, 2015) This elevated energy use in modern office buildings results from several factors: 1) lack of passive cooling strategies such as natural ventilation, replaced by the use of mechanical cooling; 2) high share of window area in building façades, causing higher solar heat gains (fully-

glazed façades being the extreme case); 3) lack of external shading, e.g. in the form of brise soleil or overhangs. (Lima, 2010; Minku, 2005; Tamanini Junior and Ghisi, 2015; Veloso et al., 2017) Moreover, temperature increase due to global warming will likely increase the cooling needs even further. In Belém, a city in northern Brazil, the use of passive cooling strategies will become almost completely inviable in the next few decades. (Invidiata and Ghisi, 2016)

As can be seen, the electricity consumption in Brazilian office buildings has been increasing for the past years and is likely to increase in the future. The energy system has started to incorporate more fossil fuel energy in its electricity mix in order to meet the growing demand. Accordingly, greenhouse gas emissions from electricity production are increasing at an even faster rate. The negative impact of GHG emissions could be partly limited by reductions of energy use in office buildings. Indeed, it has been shown that cost-effective energy efficiency in commercial buildings is a significant part of greenhouse gas mitigation potential in the Brazilian building sector. (de Melo et al., 2013; McKinsey & Company, 2009) However, surveys among commercial building owners have shown that they are often not aware of the energy reduction potential and they believe the energy consumption cannot be reduced. (Eletrobrás, 2008) Consequently, a number of studies have tried to characterize the current energy consumption level of office buildings and to investigate how the energy demand can be moderated through informed building design and the use of energy-efficient devices. A whole subchapter of this work is devoted to the review of these energy-related studies (see 2.2). These research efforts have contributed to the creation of national initiatives aimed at increasing the awareness of the most beneficial energy-efficiency strategies, with some national programs focused on office buildings in particular. (CBCS, n.d.; PBE, n.d.; PROCEL, n.d.; ProjetEEE, 2020) Despite these activities, a lot remains to be done. A recent report from International Energy Agency identified that reduction of cooling loads in buildings and development of building codes, are two immediate priorities for Brazil to address the building sector growth, its energy consumption, and carbon footprint. (IEA, 2019a)

However, energy efficiency is not the only concern of the Brazilian building industry, as the increasing rate of construction has a range of other impacts as well. Another consequence is the need for more construction materials, particularly concrete and steel. Concrete-based construction is typical for Latin America as a whole, which has one of the highest shares of concrete in building material composition among all world regions. (IEA, 2018) Obviously, the construction practices are not uniform across Brazil: being a country of a considerable size, there are many regional differences, also in the construction sector. The share of residential buildings with wooden walls may serve as an example: there are almost no such buildings in the south-east of the country (states of São Paulo, Minas Gerais, and Rio de Janeiro), while almost half of all residential buildings are made this way in the state of Acre in the west of the country, amidst the Amazon rainforest. (IBGE, 2019b) Nonetheless, buildings made of reinforced concrete and brick masonry with mortar bonding are predominant in most of the country, including the most populated regions.

The need for buildings and infrastructure drives the consumption of steel and cement. As a result, the energy use for production of these two materials shows a growing trend, even though the energy consumption per unit material has been systematically declining since 1970, due to efforts of the Brazilian industry. (MME, 2019) Additionally, the production of construction materials becomes a burden for the environment and the livelihoods of local people. Production of steel requires iron ore and charcoal, both of which induce deforestation of the native Amazon forest. (Sonter et al., 2017) Reis et al. (2019)

performed a material flow analysis study of Brazilian concrete and mortar, and found that national statistical data covers only 10% of all sand needed for concrete and mortar production, with the remaining 90% coming from illegal sand mining activities and small establishments. (Reis et al., 2019) Regulations are unlikely to be effectively imposed on such small-scale and illegal mining activities and thus the potential impact on ecosystems is high.

The production of construction materials leads to numerous negative impacts and the situation is aggravated by the way these materials are managed. In the past years, a number of studies investigated material waste in Brazilian construction sites. (Agopyan et al., 1998; de Souza et al., 2013; Formoso et al., 2002) Clearly, high material waste is a loss for the construction company itself, so there are companies in which the losses are extremely low, but there remain some in which the losses are so high they may even threaten the future existence of the company. (de Souza et al., 2013) Even the most recent study by Reis et al. (2019) shows low material use efficiency: 53% for concrete and 34% for mortar. The authors point out that industrialization of concrete and mortar production could limit the waste rates, reduce CO<sub>2</sub> emissions and even limit informal mining of coarse aggregates used for the production of these materials. (Reis et al., 2013; Formoso et al., 2002)

The Brazilian waste management system also awaits improvements, as landfills are currently the main waste disposal method and recycling remains marginal. (OECD, 2015) Brasileiro et al. (2015) performed a literature review of construction and demolition waste (CDW) treatment opportunities in Brazil and found out that majority of CDW waste is discarded in landfills even though the construction sector has a large potential for absorbing recycled CDW. The authors note that collection issues and high initial investment halt the progress of the recycling sector. The market for recycled materials is limited since potential clients are discouraged by the lack of norms that would control the quality of the secondary material, characterized by high variability. (Brasileiro et al., 2015) Low-income clients are the main customer group of demolition products, as such second-hand building materials are often the only ones they can afford. (da Rocha and Sattler, 2009)

The construction sector in Brazil causes progressing deterioration of the environment. Unfortunately, even the country's capital shows weak environmental performance as the existing regulation is not properly reinforced. (Costa et al., 2018) As a result, there is a visibly higher willingness to pay for green buildings compared to developed countries, which is further reinforced by the relative scarcity of such properties. (Costa et al., 2018) Unfortunately, this trend leads to marketing strategies of advertising office buildings as sustainable, even when sustainability is not addressed in an objective way. (Fossati et al., 2008) Sustainability aspects of these buildings should be fully understood so that a legal basis for voluntary and obligatory programs could be created and enforced.

The improvement of office building design, construction and operation could decrease the environmental pressure by reducing the demand of energy and materials, while providing other co-benefits for the society.

The next subsections will cover the recent developments in understanding the energy and material consumption patterns of Brazilian office buildings, and the resulting GHG emissions.

## 2.2 Energy characteristics of Brazilian office buildings

Operational energy use has been proven to be the biggest contributor to environmental impacts during the life cycle of typical buildings. (Ramesh et al., 2010) At the same time, energy performance of a building results from an interaction of multiple elements, some easier to control than others. Yoshino et al. (2017) identified six main factors influencing energy use: 1) climate; 2) building envelope; 3) building systems; 4) building operation and maintenance; 5) occupant behavior; 6) indoor environmental quality. These factors often interact.

*The building envelope*, made of external walls, windows, roof and ground-facing floor, separates the building from the surrounding world. Thermal parameters of the building envelope determine the extent to which the building is influenced by *climatic conditions* such as the outdoor temperature, humidity, and solar gains. In warm climates, window-to-wall ratio (WWR) strongly influences the solar gains and thus the cooling loads.

*Building systems* are often referred to as HVAC systems, as they control heating, ventilation, and air conditioning. Proper *indoor environmental quality* is strongly dependent on the efficient operation of these systems. Their energy performance is influenced by the efficiency. Cooling system efficiency is often given as the coefficient of performance (COP), which is the ratio of useful cooling output to work input. The higher the COP value, the more efficient the system. Building systems may also include plumbing, life safety and others. Mechanical air-conditioning (AC) system may be substituted by passive cooling, known as natural ventilation (NV). Some buildings employ a hybrid strategy, known as a mixed-mode (MM) system.

The energy use is also influenced by parameters related to *building operation*, the most important being the desired indoor temperature levels (temperature setpoints), the fresh air inflow needed for ventilation, and the operational schedule. Building operational characteristics also cover the intensity of internal heat gains from building's occupants, lighting, and electric equipment. These heat gains can lower energy use during the heating season, but they increase it during the cooling season.

Last but not least, *occupant behavior* has been found to strongly affect energy performance, for example causing increased heat exchange with the surroundings by opening windows during heating or cooling regimes of the HVAC systems. The effect of behavioral impacts can be surprisingly big. Up to 10 times difference in energy use has been observed worldwide in office buildings with the same climatic conditions, building functions, occupancy levels and indoor environmental quality. (IPCC, 2014)

All these factors influence the energy use in buildings, although some have been more investigated than others. The following paragraphs describe the findings of researchers investigating energy use of Brazilian office buildings, with research focused on one or more of these factors.

The first studies on Brazilian office buildings typically focused on data collection and statistical analysis of building features. (Carlo, 2008; Fialho, 2007; Lamberts et al., 2006; Minku, 2005) Clearly, before tools such as Google Earth were available, even the investigation of external building features required on-site observations and was time consuming.

Minku (2005) was among the first ones to note the raising popularity of office buildings with highly glazed façades and no appropriate external shading, resulting in poor energy

performance. Her study on office buildings in Florianópolis was used as a benchmark for another study by Tamanini Junior and Ghisi (2015) who performed a similar analysis ten years later, and concluded that building designers continue to have little concern for energy efficiency because such an energy-intensive building type became even more popular during this 10-year period. Many other researchers also point to the low energy efficiency of this inefficient office building design with high WWR, strongly dependent on artificial ventilation and lighting. (Carvalho et al., 2010; Lima, 2010; Neves and Marques, 2017; Veloso et al., 2017) Low solar heat gain coefficient (SHGC) can improve the energy performance of this archetype (Carvalho et al., 2010; Lima, 2010; Neves and Marques, 2017), and the visible correlation between fully air-conditioned buildings and low SHGC value observed by some researchers (Veloso, 2017, p. 100; Veloso et al., 2017, Fig. 6) suggests that building designers already exploit this energy efficiency strategy. Interestingly, one author notes that this highly glazed building type is mostly valued for aesthetics, but the occupants are often not satisfied with the performance: fully glazed façade type often makes it necessary to use internal shading to reduce glare issues, imposing the use of artificial lighting and depriving the occupants of visual contact with the outside world. (Lima, 2010)

Low thermal performance of office buildings was also indicated by Lamberts et al. (2006), who used the work of Minku (2005) and other students to analyze features of office buildings spread around the country. The authors pointed out that some characteristics were found to be representative in all regions such as rectangular shape, walls made of ceramic brick masonry, roof made of a slab with either a waterproof cover or ceramic tiles. The study claimed that such a homogeneity is an indication of inefficient design as the office buildings are located in different climates. However, the investigated sample was very small so only limited conclusions can be drawn from the study.

Santana (2006) performed one of the first studies investigating the influence of design parameters on energy performance of Brazilian office buildings. The author found that WWR had the biggest influence on the energy use, followed by wall reflectance, shading and HVAC efficiency. Veloso (2017) performed a multivariate regression analysis and confirmed the importance of WWR and wall reflectance on energy use intensity (EUI). Additionally, Veloso (2017) observed that EUI is highly influenced by SHGC and building ventilation mode, i.e. fully air-conditioned, mixed-mode and fully naturally ventilated. However, due to data availability issues, the regression performed by Veloso (2017) did not include the energy use for lighting and equipment or HVAC system efficiency.

Some studies focus on a particular building type. The Brazilian bank building stock was investigated by Borgstein and Lamberts (2014) and Wong et al. (2019). Borgstein and Lamberts (2014) focused on the influence of climate on EUI and found out that climate zones, as defined by the Brazilian standard NBR 15220 (ABNT, 2003), cannot be used as an adequate correction factor for EUI. The authors show by means of energy simulations that EUI can be better predicted using climatic conditions characterized using cooling degree hours (CDH), especially when calculated using wet-bulb temperature.

Wong et al. (2019) performed an analysis of the bank building stock in Curitiba. By means of statistical methods, they discovered that WWR and occupant density were the two parameters that stand out when differentiating between low and medium-to-high energy consumers.

Another important study was done by Lamberts et al. (2015), who created an energy consumption benchmark for a building archetype called a corporate office: high-rise, with

floor area more than 700 m<sup>2</sup> on each floor, with both private and common areas, and close to 100% of fully air-conditioned floor space. The final benchmark of energy use intensity (EUI) includes considerations of factors strongly influencing the energy use but varying from one building to another, such as such as climatic conditions, number of elevators, occupancy density, parking area, and the existence of other energy uses (data processing center, kitchen, emergency power source). In this truly innovative study, the authors gather enough empirical and experimental data to conclude that the cooling system type is not a strong predictor for energy use, as long as we compare only fully air-conditioned buildings. This project gave rise to a benchmarking platform available online. (CBCS, 2016)

An important contribution to characterizing Brazilian office building archetypes was made by Alves et al. (2017), who proposed a methodology for estimating office building EUI, using a case study of high-rise office building stock in Belo Horizonte. This innovative method involved unique methods such as review of land use legislation, use of a building database based on land tax information, and analysis of external building features using Google Earth. Interestingly, external building features were even used to determine the AC system type. Based on all this data, Alves et al. (2017) created three office building archetypes, characterized by different construction period, ventilation mode and other design and operational parameters. This work formed the basis for another journal article which estimated the possible energy savings related to retrofitting of the older office building archetypes. (Alves et al., 2018)

A large group of studies focused on the EUI reduction through passive cooling strategies: natural ventilation (Figueiredo, 2007; Marcondes, 2010), shading (Capistrano, 2008; Lima, 2012), thermal inertia (Brito, 2015; Ramos, 2010), and a combination of different passive strategies (Maciel, 2002). A comprehensive list of passive strategies can be found on a ProjetEEE webpage – a project run by the Brazilian Ministry of Environment that intends to increase energy efficiency in commercial and public buildings. (ProjetEEE, 2020) Research on natural ventilation is accompanied by a wide range of studies primarily aimed at occupant comfort in naturally ventilated and mixed-mode Brazilian buildings. (André, 2019; Andreasi, 2009; Cândido, 2010; De Vecchi, 2015; Lamberts et al., 2013; Marcondes et al., 2012; Mariana, 2013; Neves et al., 2020; Rupp and Ghisi, 2017) These studies show that occupant comfort may go hand in hand with low energy use if careful design of naturally ventilated spaces is employed.

Even with such robust evidence advocating for the use of natural ventilation, fully airconditioned buildings are usually preferred, partly due to high land prices in the main urban centers, which makes the investors maximize the useful floor space in order to keep high return on investments. (Damasia Gonzalez dos Santos Oliveira, personal communication, 24 March 2020 and 1 April 2020) Indeed, naturally ventilated buildings should be designed with narrow plan depths for sufficient flow of air across interior spaces (Wood and Salib, 2013, p. 165), which can potentially result in smaller floor areas.

Even though mixed-mode buildings do not deliver the true potential of natural ventilation since a mechanical system needs to be provided anyway, they are usually the best-case scenario in real buildings. (Wood and Salib, 2013, pp. 11–12) Mixed-mode buildings are the main subject of a study by Santesso and Chvatal (2018), which explored the influence of different parameters on the performance of such systems in São Paulo-based offices. The study focused on parameters such as room shape, WWR, effective opening area in windows (the percentage of window area that can be opened for ventilation) and cooling setpoint. The authors show that MM buildings offer substantial energy savings over AC

buildings in the climate of São Paulo. An important finding of the study shows that just one degree of difference in the cooling setpoint can cause as much as 30% difference in EUI. Possible energy savings are smaller in case of climates warmer than São Paulo. (Brugnera, 2014)

Pereira (2019) investigated 50 real MM buildings in São Paulo to collect information on design strategies of such buildings. Based on this data, energy simulations were performed, which confirmed the previous findings of Santesso and Chvatal (2018) on the significance window opening effective area on energy use. Pereira (2019) also noted the importance of window type and window positioning in the ventilated area.

The most remarkable study on mixed-mode buildings in Brazilian reality was performed by Neves et al. (2019) who performed a full sensitivity analysis of various factors influencing energy loads in MM buildings, such as window opening effective area, solar absorptance, shading, thermal capacity, SHGC, U-value of walls, floor height, WWR, and solar orientation. The authors show through a Monte Carlo analysis that window opening effective area and SHGC are more important for buildings with limited external shading. These three parameters are also the most significant predictors for cooling loads, according to a multivariate regression analysis.

Some important conclusions can be drawn from the work presented above. Firstly, there is a highly active research community in the area of energy performance of Brazilian office buildings. This is despite of data availability issues raised by multiple researchers, especially data for real energy use values. (Alves et al., 2018; Borgstein and Lamberts, 2014; Lamberts et al., 2015; Veloso, 2017) There is also no office building database with details such as physical parameters and operational regimes, required for energy simulations. (Alves et al., 2018)

Secondly, many of the studies focus on a particular group of office buildings, such as corporate offices (Lamberts et al., 2015), bank branches (Borgstein and Lamberts, 2014; Wong et al., 2019) or mixed-mode buildings (Neves et al., 2019; Pereira, 2019; Santesso and Chvatal, 2018). The results of these studies cannot be directly applied to the whole office building stock, but their results could be combined to get a fuller picture of the stock.

Furthermore, relatively little energy research is performed on a national scale, limited to the work by Borgstein and Lamberts (2014), and Lamberts et al. (2015), both focused on such a subgroup of office buildings. This scarcity can be partly explained by low availability of energy use values in real buildings.

Lastly, the presented research suggests that shading, SHGC and WWR are among the most influential parameters for energy performance of office buildings. As shown by Neves et al. (2019), the importance of these parameters changes depending on the baseline conditions but such general trends can be observed in any case. Window opening effective area is another important factor for energy performance, applicable only in case of mixed-mode buildings. (Neves et al., 2019; Pereira, 2019; Santesso and Chvatal, 2018)

## 2.3 Material characteristics of Brazilian office buildings

As described in the previous section, multiple researchers analyzed features of Brazilian office buildings that are important to their energy performance. The typical materials included in such analyses are materials included in the building envelope and floors, such as concrete, bricks, plaster cement, window glass, roof covering. However, there are

multiple materials that have no direct impact on energy use of the building while having a significant influence on material use, and consequently, on emission performance.

Unfortunately, there are no studies investigating the material composition of Brazilian office buildings. Such studies are not available even for residential buildings, which have typically much better coverage in building research. However, there was one study which also required detailed data on material composition: a material flow analysis of residential building stock in Rio de Janeiro, performed by Condeixa et al. (2017). The authors based their material composition modeling on data from the Syndicate of Construction Industry (Sinduscon-MG, 2007), which describes typical design standards for construction projects, as described in NBR 12721, which is a Brazilian standard on construction cost assessment. (ABNT, 2005) These building designs are typical and representative of buildings in Brazil.

Condeixa et al. (2017) mention also another source of material data, so-called TCPO guide which presents Brazilian averages of material and equipment needs for typical activities during construction and demolition of buildings. (PINI, 2010)

The documents from ABNT (2005), Sinduscon-MG (2007) and PINI (2010) can serve as a basis for material composition data for Brazilian construction projects. Additionally, ABNT (2005) and Sinduscon-MG (2007) include some commercial building archetypes: 8-floor office with open-plan layout, 8-floor office with cellular layout, and 16-floor office with cellular layout. These three commercial buildings are shown in two construction standards: normal and high standard.

Furthermore, material data on Brazilian buildings could also be sourced from life cycle inventories of life cycle assessment studies. However, as will be presented in the next section, there are no such studies on whole office buildings in the Brazilian context.

## 2.4 Emission characteristics of Brazilian office buildings

Greenhouse gas emissions associated with buildings can be calculated using EN 15978. (CEN, 2012) Although this is a European standard, it is used in countries outside of Europe, including Brazil. (Gomes et al., 2018; Morales et al., 2019; Rohden and Garcez, 2018) The standard offers a framework for life cycle assessment (LCA) of buildings and building products, taking into account four life cycle stages of a building: construction, use, demolition, and possible benefits outside of the building itself, related to material reuse, recycling, and recovery.

Emissions associated with Brazilian office buildings are not well understood, as there are few national studies focused on office buildings (Brugnera, 2018; Najjar et al., 2019; Taborianski and Prado, 2012). Furthermore, none of them concerns a whole building LCA with a full analysis of all life cycle stages of the building. However, there are multiple LCA studies investigating GHG emissions of Brazilian buildings, components, and materials. This work is analyzed in the following paragraphs in order to understand the impacts of the Brazilian construction industry. Many of these studies are not specific to offices or even to commercial buildings – however, the differences within the Brazilian construction sector are still likely to be smaller than the differences between office buildings located in different countries.

As it turns out, whole building LCA studies are extremely limited in Brazil. (Gomes et al., 2018) The available studies investigate social housing (Morales et al., 2019; Paulsen and Sposto, 2013), residential buildings (Evangelista et al., 2018) and university buildings (Gomes et al., 2018). Even though these building types are distinct from office buildings,

these whole building LCA studies are nonetheless analyzed in order to understand the most important takeaways.

Paulsen and Sposto (2013) were among the first ones to perform a whole building LCA study in Brazil. The operational energy use was estimated based on statistical data, assuming that cooling loads were zero due to lack of cooling devices. The only estimated environmental impact was energy – embodied energy for material-related stages, and operational energy.

Morales et al. (2019) performed another study of social housing. Again, the use of AC was disregarded. The authors analyzed multiple environmental impacts, including climate change impact measured using global warming potential (GWP) for a time horizon of 100 years, denoted as GWP100. The results show the relative importance of operational energy use for GWP impacts while the material GWP impacts were dominated by concrete, steel and cement mortar. An important contribution of this study was a comparison of material impacts depending on life cycle inventory (LCI) data adopted: global data resulted in up to two times higher GWP impact than regionalized data. Regionalization was performed by adapting the available data to the Brazilian electricity grid and local production processes.

Evangelista et al. (2018) investigated 4 types of Brazilian residential buildings in a whole building LCA study. The GWP impacts differed with the building type and the highest GWP impact was associated with a low standard single-family housing. Again, material-related emissions were dominated by concrete and steel.

Gomes et al. (2018) performed an LCA study of a university building. This is the only whole building Brazilian study that incorporated energy simulation to determine operational energy use. A photovoltaic array was included for on-site energy generation in two modeled scenarios: net zero energy (NZE) and energy positive (E+) building. One of the conclusions is that replacement has a strong influence on the LCA results while there is high uncertainty related to the service life of building components. The authors point to modeling issues associated with the lack of national material LCI databases, forcing LCA practitioners to mix LCI databases. Stage A (pre-use phase) turned out to be the most emission-intensive, which is a typical result for NZE and E+ buildings. Emissions associated with stage C (end-of-life phase) were practically negligible.

A number of studies investigated material emissions related to Brazilian buildings. The studies show high contribution of the main structure and wall masonry to GWP impacts (Medeiros et al., 2018), and a significant influence from production of just a few materials: concrete, steel, and cement (Saade et al., 2014; Silva, 2013). Tavares (2006, pp. 144, 151) found out that paint has a relatively high GWP impact considering total material emissions (including replacements), preceded only by cement in one of the investigated cases. Saade et al. (2014) observed the impact of different concrete and cement types on the LCA results.

Material emissions were also calculated by Najjar et al. (2019), who used a multi-story office building as a case study to show the advantages of LCA and building information modeling (BIM) integration. Najjar et al. (2019) modeled two alternatives: a concrete construction using typical building materials (concrete, brick, ceramics, wooden windows) and a steel construction using drywall and curtain wall systems with aluminum window frames. The climate change impacts of the concrete building were dominated by cement and ceramics while impacts of the steel construction – by aluminum and steel. Najjar et al. published another similar study in 2017. (Najjar et al., 2017)

Moreover, Ecoinvent 3.6 database includes three datasets of Brazilian hotels. (Santa Rosa Rocha, 2019a, 2019b, 2019c) The datasets are modeled to include the whole construction phase and material replacement, which is equivalent to modules A1-A4 and B4 according to EN 15978. (CEN, 2012) Taking into account the building area (including ancillary floor area) and building lifetime, the GWP impacts associated with these hotels range from 3.9 to 10.5 kg  $CO_2$ -eq/m<sup>2</sup>/year.

There are also authors who created life cycle inventories of particular Brazilian construction materials: cement and concrete (Silva et al., 2018), concrete blocks (John et al., n.d.), ceramic bricks (Soares and Pereira, 2004) and particleboard (Silva, 2012).

The Brazilian-based LCA studies investigated also particular building elements: the main structure (Bento et al., 2013; Bento, 2017; Rohden and Garcez, 2018), façades (Brugnera, 2018; Taborianski and Prado, 2012), masonry walls (Condeixa et al., 2014; Morales et al., 2020), and roof (Souza et al., 2015). This work is briefly described in the following paragraphs.

The studies on the main building structure were primarily focused on concrete type, as it is the predominant material used in Brazilian building structures. Specifically, the authors investigate structures with concrete of different compressive strength values. The least emission-intensive systems included concrete C40 (Bento, 2017), concrete C35 (Bento et al., 2013) or a mixture of all concrete types from C25 to C50, contrasted with the use of concrete C25 only (Rohden and Garcez, 2018).

The least emission-intensive façade system for office buildings is a brick wall with mortar, which is the conventional building style in Brazil, while structural glazing (also known as a curtain wall system) with colorless glass is the most emission-intensive. (Brugnera, 2018, p. 161; Taborianski and Prado, 2012) Changing the glass to reflective significantly reduces the emissions. (Taborianski and Prado, 2012) The work shows that aluminum causes the highest emission levels among all façade materials (Brugnera, 2018, p. 161; Taborianski and Prado, 2012), preceded only by aluminum composite panels (ACM) (Taborianski and Prado, 2012).

Condeixa et al. (2014) performed an LCA study of masonry walls. The authors estimated the amount of material waste and distances of transportation from material suppliers to the site, and from the site to landfill. Condeixa et al. (2014) find high impacts caused by wasted material and suggest that solutions could include the improvement of workforce qualifications, construction planning and waste management.

Morales et al. (2020) investigated masonry walls with a focus on replacement of paint and mortar. The authors discovered large impact differences depending on the choice of service lifetime for paint and mortar. It was also shown that regionalized LCI data shows lower impacts than global data.

Finally, an LCA study on roof systems done by Souza et al. (2015) concludes that ceramic tile system is associated with lower GHG emissions than concrete tile system, mainly due to high temperatures used during cement calcination.

The presented LCA research shows a growing interest in environmental assessment studies of the Brazilian construction industry. Many researchers point to representativeness issues related to the lack of national LCI datasets. (Gomes et al., 2018; Morales et al., 2019; Souza et al., 2015) Some authors use data regionalization to adapt foreign system processes to Brazilian context. (Morales et al., 2019, 2020) However, the persistent effort

of Brazilian LCA researchers allowed to create national system processes for materials such as cement and concrete, available in Ecoinvent v3.6. (Silva et al., 2018)

## 2.5 Sensitivity analysis in building research

Energy consumption and environmental impacts of buildings are subject to changes depending on the parameter choice. This fact has made multiple researchers explore this topic. The following paragraphs describe the current state of knowledge on the importance of different factors on the energy and emission performance of buildings. If available, studies on office buildings were preferred.

Lam and Hui (1996) performed a sensitivity analysis of energy use in office buildings in Hong Kong. The authors found out that cooling setpoint, cooling efficiency parameters (chiller COP) and occupancy density are the most impactful factors. Shading and WWR were among the least impactful factors.

Korolija et al. (2013) investigated the office building stock in UK. Although the authors did not perform a thorough sensitivity analysis, they created building archetypes using model parametrization, which allowed them to create a whole array of buildings that represent the majority of the existing building stock. Therefore, this study shows that using Monte Carlo or other sampling methods to represent variability in building stock is a valid method for energy performance or carbon emission analysis.

As already described in the previous sections (see 2.2), Santana (2006) and Veloso (2017) analyzed parameters influencing energy use in Brazilian office buildings. Using multivariate regression analysis, Santana (2006) concluded that EUI is highly dependent on WWR, wall reflectance, shading, and HVAC efficiency while Veloso (2017) underlined the importance of SHGC and the building cooling strategy.

Another study introduced in the energy research section is a remarkable study by Neves et al. (2019), which includes a full sensitivity analysis of energy loads in Brazilian mixed-mode buildings. The study used sensitivity analysis methods such as OAT (one-at-a-time), Morris method and Monte Carlo with a multivariate regression to show that the most important parameters are window opening effective area, SHGC and shading.

As for the emission performance of buildings, Chay et al. (2012) analyzed  $CO_2$  emissions associated with the superstructure of a high-rise office building in Hong Kong, with the study scope covering material production, transport, assembly and replacement. The modeled building had a reinforced concrete frame and floors, aluminum windows, and plasterboard wall finishes. Material emission factors (kg  $CO_2/kg$  of material) were the source of variability in the input data, derived from embodied energy intensities for materials and emission factors for electricity. The average  $CO_2$  emissions was found to be 215 kg  $CO_2/m^2$ , with the 95% confidence interval from 86 to 460 kg  $CO_2/m^2$ , assuming a lognormal curve. External walls and internal floors were the biggest contributor to this impact, followed by suspended ceilings and finishes. The authors also explore strategies for reducing the emissions, concluding that lifetime extension for the building components is the most effective option, followed by diverting construction waste to recycling.

Hoxha et al. (2014) performed another sensitivity analysis study focusing on material impacts. The authors use a single-family detached house in France as a case study to demonstrate sensitivity of the overall environmental impact on parameters such as material lifetime, material amount and elementary impact of the material production processes. It was found that on the whole building level, thermal insulation, bitumen, and

PVC are the three materials contributing the most to the uncertainty. On the material level, service life was usually the biggest source of uncertainty. For a few materials, uncertainty related to elementary impact was the most important (concrete, structural clay materials, structural wood, plaster). Material amount was the smallest source of material-level uncertainty, but it was likely influenced by the modeling choices – the material amount was only varied between -5% and +10%. Reinforced concrete variant of the building was characterized by a variation in GWP impact between 469.7 and 674.5 kg  $CO_2$ -eq/m<sup>2</sup>.

Heeren et al. (2015) explored the possible environmental impacts associated with wooden and massive (concrete-based) residential and small office buildings in the context of Switzerland. In a Monte Carlo analysis, 22 variable parameters were randomly sampled. The parameters were divided into a few categories: material (e.g. construction material), design (e.g. WWR), operation (e.g. occupancy) and exogenic parameters (climate, electricity mix). The results show that the massive variant (more relevant to this study as Brazilian buildings are mostly concrete-based) shows the biggest correlation between GWP impacts and the following parameters (in descending order): electricity mix, ventilation rate, thermal generator type, construction material, material service life.

Häfliger et al. (2017) investigated sensitivity of impacts related to material modeling choices. Again, Switzerland was used as a case study. In particular, the authors focused on the sensitivity related to database choices, system boundary definitions and scenarios for replacement of building materials. Wood and insulation turned out to be among the factors with the highest impact variation depending on the database choice. As for the replacement scenarios, insulation and windows and doors have a high impact variation, while at the same time they make up a large share of total impacts. The total GWP impact uncertainty caused by different replacement scenarios depends on the building type and its building lifetime, but it ranges between 10 and 20% of the overall impact.

Finally, a previously mentioned Brazilian study on masonry walls performed by Morales et al. (2020) confirmed the findings by Häfliger et al. (2017) suggesting a big influence of material service life on the final GWP values.

The work on sensitivity analysis in building research presented above allows to draw some important conclusions. First of all, one should be careful when comparing the results of energy-related sensitivity analyses performed in locations with different climatic conditions. Lam and Hui (1996), based in Hong Kong, found that shading is the least important while Santana (2006) and Neves et al. (2019), based in Brazil, concluded that shading is among the most important parameters.

Multiple researchers have focused on the environmental impacts related with materials, which are not as location dependent as studies involving operational energy use. It was found that material service life can have a high contribution to variance for GWP values. (Häfliger et al., 2017; Hoxha et al., 2014; Morales et al., 2020)

Lastly, the study by Heeren et al. (2015) was the only one which analyzed the impact associated with all the life cycle stages, including the operational energy use. In this work, the authors show that electricity mix shows the strongest correlation with GWP impact.

This chapter summarized the available literature, showing the current state of knowledge on Brazilian office buildings: their energy and material requirements and the resulting emissions. As it was shown, there are no studies on emission performance of these buildings, showing an important research gap that remains to be addressed. The next chapter will describe the methodology adopted to analyze the carbon footprint of Brazilian office buildings and perform a sensitivity analysis of the results.

# 3 Methodology

This chapter describes the steps taken to address the research questions of this study. Literature review presented in the previous chapter serves as a starting point for a model, whose main purpose is to assess the possible range of carbon footprint values of Brazilian office buildings.

The framework used in this work consists of several steps, which is illustrated in the form of a schematic representation in Figure 2. Firstly, office building archetype is defined. Physical characteristics of the modeled buildings are specified, based on the data collected by means of the literature review. The parameters considered critical for building performance are characterized in terms of their typical range. Some of them are chosen to resemble those typically found in real buildings and others are chosen to investigate possible improvement options. Later, these parameters are being altered as a means of exploring a wider range of carbon footprint values.



#### Figure 2 Schematic representation of the methodology.

The second step involves a description of the sample selection method. The variable parameters chosen in the previous step open a wide range of possible combinations, and it is unfeasible to investigate them all due to computational constraints. Therefore, sample selection methods are required to sufficiently cover the parameter space in a way that is not too computationally intensive. As a result, a sample with the selected parameter combinations is chosen, representing multiple buildings.

As the next step, the chosen sample of buildings is modeled in building energy simulation software. At this point, some additional modeling assumptions need to be made, partly as a means of model simplification in aspects that are unlikely to have a large result on the carbon footprint calculations.

Further, material modeling is performed in order to quantify material flows associated with the modeled buildings. A part of the data is extracted from the energy simulation software while the rest is applied based on the collected building archetype data. This procedure makes it possible to investigate the interaction between energy and materials, so evident for example in case of a building envelope.

Next, GHG emission modeling is done using the quantified flows of energy and materials from the previous stages. The magnitude of these flows is used to calculate the carbon footprint associated with the modeled buildings. The impact is divided into building life cycle stages, with accordance to EN 15978 standard (CEN, 2012).

Lastly, a sensitivity analysis of the results is performed. The life cycle assessment done in the previous stage results in a whole range of climate change impact values for the buildings. These values are analyzed by means of a multiple linear regression in order to understand the sensitivity of the calculated carbon footprint to changes in the variable parameters.

The following sections describe each of these steps in more detail.

### 3.1 Office building archetype definition

The first step of model development requires definition of building archetypes, which includes specification of all the parameters needed as inputs in building energy software, such as climate, building geometry and construction materials, windows and shading, HVAC system, internal heat gains, and others. The archetype definition includes also full material composition description, needed for material and emission modeling.

The following subsections describe the assumptions made with regards to these parameters. While most of the parameters are fixed, some of them are considered variable as they have a particular significance for energy or emission performance. A range of typical values is assigned to each such parameter.

#### 3.1.1 Climate

As has been shown in numerous studies, climate is one of the most important predictors of energy use in buildings, which also applies to Brazilian office buildings. (Borgstein and Lamberts, 2014; Lamberts et al., 2015) Thus, climate is subject to alterations during the simulation stage. The simulated building is located in one of the 12 biggest Brazilian cities: São Paulo, Rio de Janeiro, Brasília, Salvador, Fortaleza, Belo Horizonte, Manaus, Curitiba, Recife, Goiânia, Belém, and Porto Alegre. These cities were depicted in Figure 1, which also shows that such a selection of cities ensures there are at least two cities in a given region. As explained in section 2.1, office floor area can be approximated using employment statistics such as IBGE (2019a). The 12 chosen cities are the biggest cities in their 12 respective states (IBGE, 2011), and these states employ almost 85% of the total number of workers employed in office-related professions such as IT, communication, finance, real estate, professional and administrative activities. (IBGE, 2019a) Therefore, these cities are likely to represent the great majority of office floor area in the country.

#### 3.1.2 Geometry and construction materials

The archetypes are based on commercial building types specified in the Brazilian national standard NBR 12721 (ABNT, 2005), further described by the Syndicate of Construction Industry (SINDUSCON) (Sinduscon-MG, 2007). NBR 12721 lists multiple building types, including residential and commercial ones, specifying their main characteristics and material composition, while Sinduscon-MG (2007) gives additional information on each of these buildings, such as floor plans. These building types were created for construction cost assessment purposes and the data used for their creation represents typical construction practices. Thus, building archetypes built upon this data should be a good representation of typical Brazilian buildings.

Three archetypes were considered in this work, all of them based on NBR 12721 and their commercial building types built in normal standard. NBR 12721 distinguishes also high-standard constructions, but relatively small differences between the two standards exist. Three office building archetypes were considered, and they will be referred to as OPL-8,

CL-8, and CL-16. The main information about them can be found in Table 1. The material data of the archetypes is listed in Appendix 1.

It must be noted that NBR 12721 lists the material data in the form of components, which needed to be converted to the amount of a given material type, for the purpose of this work. Some assumptions needed to be made in the process. Additionally, some components listed in NBR 12721 have been excluded from this work, such as door handles, circuit breakers, pressure valves, steel pipes. The amounts of these components were insignificant so this simplification should not influence the results of the study.

	Archetype 1 (OPL-8)	Archetype 2 (CL-8)	Archetype 3 (CL-16)
Reference in NBR 12721	CAL-8	CSL-8	CSL-16
Standard	Normal	Normal	Normal
Floor layout	Open plan	Cellular	Cellular
Floors	8	8	16
Number of elevators	2	2	3

able 1 Three office	building archetype	s and their main	characteristics.

All the considered buildings have a rectangular form, with a footprint of 20 meters by 30 meters, and the long axis aligned with the east-west axis. Given such footprint, each floor has 600 m<sup>2</sup> of floor area, which is similar to the floor size of the building types described in NBR 12721. The buildings have either 8 floors (OPL-8 and CL-8) or 16 floors (CL-16). Depending on the city, the most common office building height falls within the range of 10-15 floors. (Alves, 2017; Fialho, 2007; Minku, 2005; Neves et al., 2019) This way, buildings of 8 and 16 floors can be considered typical, as they represent the two sides of this spectrum. The height of each floor is 2.8 meters.

Based on the collected material intensity data (Appendix 1), the archetype buildings are considered to have their main structure made of reinforced concrete. Both internal and external walls are made of brick masonry with mortar bonding. The bricks are 9 cm wide, with 2.5 cm of plaster on both sides of the wall, which is a structure typically found in Brazilian construction. (Morishita et al., 2011) Such wall construction is used for internal walls in all of the archetypes, and external walls for 8-floor archetypes (OPL-8 and CL-8). The 16-floor archetype (CL-16) has the same type of external wall construction but with a brick of 13.5 cm thickness, another brick size common in Brazil. (Morishita et al., 2011) Internal walls were modeled so that the total amount of brick intensity agrees with the data from NBR 12721: 21.6 m<sup>2</sup> and 146.4 m<sup>2</sup> of internal walls per floor for the open plan layout (OPL-8) and cellular layout (CL-8 and CL-16), respectively. These additional walls influence the energy simulation because their mass increases thermal inertia of the building.

The roof is covered by fiber cement roof tile of 6 mm thickness, as reported in NBR 12721. All the internal floors and the roof include a 10 cm-thick concrete slab, while the external floor includes a 20 cm-thick concrete slab to account for the thermal mass associated with foundations. Additionally, there is 6 cm of air space between the fiber cement roof tile and the concrete slab. (Morishita et al., 2011) On each floor, there is a 5 mm carpet tile and a 125 mm-thick acoustic ceiling with an additional ceiling air space. These two materials

were not present in the original dataset from NBR 12721 but were added because such layers decrease the extent to which the thermal mass of a concrete floor slab can be utilized, thus influencing energy performance of the building.

The materials mentioned above are used as inputs to the energy simulation. However, most of the materials listed in NBR 12721 do not have any influence on operational energy use of the building, so they are considered only at the material modeling stage of model development (see 3.4).

#### 3.1.3 Windows and shading

Various studies have suggested the importance of window-to-wall ratio (WWR) for energy performance of buildings. (Lima, 2010; Santana, 2006; Wong et al., 2019) Therefore, WWR is also an important factor for emission performance, considering that operational energy often constitutes a majority of the carbon footprint of a building. (Ramesh et al., 2010) This parameter may take values of 30%, 50%, 70%, or 90%, which represents a whole range of office buildings found in Brazil: from low WWR values typically found in mixed-mode buildings (Neves et al., 2019) up to high WWR values typical for newer construction (Lima, 2010; Minku, 2005; Tamanini Junior and Ghisi, 2015; Veloso et al., 2017). The same window-to-wall ratio applies to all the building façades.

The glazing found in the windows has 6 mm of thickness. Various researchers point to solar heat gain coefficient (SHGC) as a parameter that can strongly influence the amount of solar heat gains inside the building, and thus the energy need for cooling. (Carvalho et al., 2010; Lima, 2010; Neves et al., 2019; Neves and Marques, 2017) Therefore, SHGC is varied during the building simulations and it takes on a value of 0.2, 0.4, 0.6 or 0.8.

Window opening effective area can be defined as the share of window that can be opened for ventilation purposes. This parameter is an important factor for cooling load of mixed-mode buildings, in which windows can be opened to benefit from natural ventilation when appropriate conditions are met. (Neves et al., 2019) On the other hand, it does not influence the energy use of fully air-conditioned buildings, in which windows are often fixed because opening of windows can be detrimental to the performance of the HVAC system. As such, window opening effective area was chosen as another variable parameter, whose value also reflects the choice of cooling strategy.

A fully AC-dependent office building is often referred to in Brazilian scientific literature as the typical modern archetype. (Carvalho et al., 2010; Lamberts et al., 2015; Lima, 2010; Veloso et al., 2017) Therefore, a window opening of 0.0 represents this cooling strategy, which can be reasonably assumed to be the most common in real Brazilian office buildings.

Mixed-mode cooling strategy is likely less common in real buildings, which is reflected in visibly smaller research coverage, even though this strategy allows for significant energy savings. (Neves et al., 2019) Therefore, it was decided to include mixed-mode cooling regime in this study in order to explore possible energy savings, which are likely to also affect emission performance. Window opening effective area is typically around 0.27 (Neves et al., 2019), but a wider range of values was adopted to explore possible improvements related with mixed-mode ventilation implementation. The parameter takes on values of 0.1, 0.3, 0.5, 0.7 or 0.9, which correspond to the range considered by Neves et al. (2019).

External shading is another parameter related to windows that was proven by many to have a large importance on energy use of the building. (Carvalho et al., 2010; Neves et

al., 2019; Santana, 2006; Veloso et al., 2017) Again, a whole range of parameters was adopted to explore possible emission savings, even though most real buildings have almost no external shading. (Lima, 2010; Minku, 2005; Neves et al., 2019; Pereira, 2019; Tamanini Junior and Ghisi, 2015)

Shading can take various forms, among which overhang is the easiest one to model so it was the shading type assumed in this study. During the simulations, shading takes on values from 0.0 (no shading) up to 1.0 (overhang of the same depth as the height of the window below), with steps of 0.2.

#### 3.1.4 HVAC system

It was decided that an ideal HVAC system will be modeled in this study: a system that meets all the load requirements but consumes no energy. The cooling systems common in Brazil can be generally divided into a few types: central AC system with fan coil units, self-contained units, split units, window units, variable refrigerant flow (VRF) systems. (Lamberts et al., 2015; Pereira, 2019) However, it was found that the system type doesn't determine energy use of the building – even though some systems are usually less energy-efficient than central AC systems, they do not require chilled water pumping through the building. (Lamberts et al., 2015) Although some research suggests that buildings with window or split units are less energy-intensive (Veloso et al., 2017), it is often a result of differences in cooling regimes. These devices are often seen in mixed-mode buildings (Pereira, 2019) and, as a consequence, they are not more energy-efficient per se, they are just less utilized. Therefore, modeling the HVAC system as an ideal system should not influence the representativeness of the results to a significant degree, while allowing for considerable time savings in data collection and modeling.

The cooling and heating loads are then translated to energy use by means of energy efficiency coefficients. In case of heating, electric heating with 100% efficiency is assumed. Efficiency of cooling is decreased in relation to theoretical values to account for energy use of the HVAC system. Doing otherwise may lead to underestimated energy consumption values which is undesirable considering its typically high contribution to emission performance of the building. Additionally, the prevalence of cooling needs in Brazil suggest that cooling efficiency is a significant factor for emission performance. As a result, a range of efficiency values was adopted, based on literature.

Brazilian study performed by Inmetro (2017) have shown that split and window units available on the market are generally characterized by COP values of 2.8 to 3.3 W/W. On the other hand, Lamberts et al. (2015) found that a typical value of COP in corporate offices (dominated by central AC systems) is 4.8 W/W. Consequently, cooling efficiency values adopted in this study are in range of 2 to 4, which is lower than the COP values found in the literature to account for energy use of the HVAC system.

It was assumed that no energy source other than electricity was used. This is supported by figures on energy consumption by Brazilian commercial buildings: electricity was around 92% of the total energy use in 2018. (MME, 2019) Therefore, the exclusion of non-electric energy sources is unlikely to have a significant impact on the results.

Temperature setpoints are another category of parameters that are significant to the final energy consumption of the building. In Brazil, cooling setpoint has a particular meaning as it determines the total cooling load: the lower the cooling setpoint, the higher the load. Cooling setpoint is another variable parameter, changed between 22 and 25°C, according to values reported by Lamberts et al. (2015). Heating setpoint is fixed at 20°C. It is

assumed that the HVAC system can be activated only during working hours, i.e. between 6 AM and 6 PM on weekdays.

Regarding fresh air intake, airflow rate of 7.5 liter/s/person was assumed, which is the ventilation rate recommended by the Brazilian Health Surveillance Agency. (ANVISA, 2003)

#### 3.1.5 Internal heat gains

Internal heat gains include lighting, equipment, and occupants, all characterized by intensity (occupancy rate in case of people) and schedule. The same operating schedule was used for all internal heat gains: they are assumed to have their maximum value during the working hours (from 6 AM to 6 PM on weekdays), and they are assumed to be zero otherwise. This results in a binary (0-1) type of schedule.

The power intensity of lighting and equipment was adopted as 10.5 W/m<sup>2</sup> and 14 W/m<sup>2</sup>, respectively. These values were adopted by Alves et al. (2017) for their open plan office building archetype located in Belo Horizonte, based on suggestions from the Chartered Institute of Building Services Engineers (CIBSE) (CIBSE, 2004). These values are also in line with the energy benchmark for Brazilian corporate office buildings, developed by Lamberts et al. (2015).

According to the NBR 16401-3 standard, office buildings with medium occupant density have 14 people per 100 m<sup>2</sup> of floor space. (ABNT, 2008) However, around 20% of floor area in office buildings is used for purposes other than workplaces. (stairs, elevators, corridors, technical support etc.) (Alves et al., 2017) Accounting for this, the adopted occupant density is 11 people/100 m<sup>2</sup>.

#### 3.1.6 Lifetime

Building lifetime can potentially have a significant influence on the carbon footprint (Häfliger et al., 2017), especially in concrete buildings (Heeren et al., 2015). At the same time, there are no studies on lifetime of Brazilian buildings. Brazilian standard NBR 15575-1 defines the main structure lifetime as minimum 50 years. (ABNT, 2013) However, Brazilian office buildings are normally concrete-based structures, which are known for their durability, so lifetime of 100 years could also be possible. Therefore, building lifetime value was chosen to be 50, 75 or 100 years.

However, some building components cannot be used throughout the whole building lifetime and thus they need replacement. Brazilian standard NBR 15575-1 defines building element lifetimes, distinguishing between minimum values and recommended ones, taking into account what is technically possible with today's technology. (ABNT, 2013) On the other hand, Building Owners and Managers Association (BOMA) issued a guidebook with typical lifetime values for different building components. (BOMA, 2010) Based on these two sources, lifetimes for different materials were adopted. The exact values can be found in Appendix 2. In many cases, the values have a wide range of possible lifetimes, so it was decided to include material lifetime as another variable parameter. The materials that last for the whole building lifetime are left unchanged, while the other materials have their lifetime values adopted as 75%, 100% or 125% of the values seen in Appendix 2. These three values are referred to as material lifetime multipliers.

#### 3.1.7 Archetype definition summary

The previous sections present a full description of Brazilian office building archetypes. Ten of these parameters are considered critical for energy and emission performance, so their

values are altered throughout the model simulation in order to represent a number of buildings with distinct features. The list of these variables and the values they may assume can be found in Table 2.

The next section describes how a sample of buildings is created to explore the possible carbon footprint values for different variable parameter combinations.

Variable	Values	Ilait	Comment	
parameter	Values	Unit		
City	São Paulo, Rio de Janeiro, Brasília, Salvador, Fortaleza, Belo Horizonte, Manaus, Curitiba, Recife, Goiânia, Belém, Porto Alegre	-	-	
Building type	OPL-8, CL-8, CL-16	-	Determines building height and material composition (incl. external wall type and internal wall area)	
WWR	30%, 50%, 70%, 90%	-	-	
SHGC	0.2, 0.4, 0.6, 0.8	-	-	
Window opening effective area	0.0, 0.1, 0.3, 0.5, 0.7, 0.9	-	Value of 0.0 represents a building with no natural ventilation (fully air-conditioned)	
Shading	0.0, 0.2, 0.4, 0.6, 0.8, 1.0	-	-	
Cooling setpoint	22, 23, 24, 25	°C	-	
Cooling efficiency	2.0, 2.5, 3.0, 3.5, 4.0	W/W	-	
Building lifetime	50, 75, 100	years	-	
Material lifetime multiplier	75%, 100%, 125%	-	-	

Table 2 Overview of the variable parameters for office building archetypes.

### 3.2 Sample choice

The Brazilian office building archetypes presented in the previous section have 10 variable parameters, which assume one of 3 to 12 possible values. In total, there is almost 4 million variable combinations possible, which makes the investigation of all possible combinations extremely unfeasible due to computational constraints. As a consequence, there is a trade-off between computation speed and accuracy. The choice of a suitable sampling method allows for satisfactory speed-accuracy results.

The sampling method chosen for this study is the Latin Hypercube Sampling (LHS). This sampling method proposed by McKay et al. (1979) divides the whole range of values of each variable into a number of sections, called strata. The total number of strata is equal to the sample size. Then, for each of the variables, just one sample is taken from each stratum. The components of the different variables are matched at random, making this a quasi-random method. Such sampling ensures that the parameter space is uniformly covered. Depending on the problem type, performance of LHS varies compared with other methods where random variables are independent and identically distributed (Owen, 2019), such as the classical Monte Carlo sampling. However, it has been shown both

theoretically and empirically that LHS cannot be much worse than conventional random sampling, while it often performs significantly better. (Chrisman, 2014; Owen, 2019)

Latin hypercube sampling was performed in Python programming language using the pyDOE library (Lee, 2013). The LHS method in pyDOE library returns values from 0 to 1 for each variable for each sample item. These values are multiplied by the total number of possibilities for a given variable, and then rounded down. This way, the resulting array clearly points to the parameter values, as listed in Table 2, that should be adopted for each sample item. Each such sample item corresponds to one building with a given combination of variable parameters' values. Such a sample selection method is equivalent to uniform discrete distribution for all the considered parameters – this fact, among others, means that the results will not be representative of the whole Brazilian office building stock.

A sample size of 1000 buildings was chosen for the study. This number is a trade-off between computational time (around 7 hours) and accuracy – a sample of 1000 corresponds to 100 model evaluations per parameter, which is the minimum value to reach model convergence, as described by Mutel et al. (2013).

### 3.3 Energy modeling

The software used for building energy simulations was EnergyPlus 9.2.0, developed by the U.S. Department of Energy's (DOE) Building Technologies Office (BTO). (DOE, n.d.) EnergyPlus requires a weather file (epw format) to run the simulation with climatic conditions of a given location. The climate of the 12 chosen Brazilian cities was simulated using weather files with a typical meteorological year (TMY) for each of these locations, based on data collected from weather stations of the national meteorological institute of Brazil (INMET) during the years 2000-2010. (Roriz, 2012) The weather files are available on the EnergyPlus webpage. (DOE, n.d.)

EnergyPlus input files were created based on the data described in the previous sections, with the use of open source code ParIDF developed by Santesso (2018). The code enables to modify values of variable parameters in EnergyPlus input files in a systematic, automated manner.

Building geometry was set up using zone multipliers – a feature allowing for convenient modeling of buildings with multiple stories, with the same floor size and internal gains. Zone multipliers allow for significant time savings while yielding accurate results. (Big Ladder Software, n.d.; Chen and Hong, 2018; Ellis and Torcellini, 2005)

Thermal properties of materials were modeled based on national-specific data from Morishita et al. (2011) and built-in EnergyPlus material datasets (ASHRAE, 2005). Internal walls were modeled using internal mass objects which allow to enter the surface area of internal walls, and do not require specific coordinates for these objects. The height of windows is adjusted according to the chosen WWR value.

HVAC system was modeled as an ideal loads system which meets the heating and cooling loads but consumes no energy. The heating and cooling loads calculated by EnergyPlus were translated to electricity use at a later stage of model development.

Mixed-mode ventilation was modeled based on Neves et al. (2019), with the help of the open source code by Santesso (2018). Multizone airflow network (AFN) objects were used to simulate window openings. The airflow through windows was modeled using AFN surface objects with the window opening factor equal to the chosen window opening effective area
(see Table 2). Energy management system (EMS) objects were used to model mixed-mode ventilation on each building floor. By means of code written through EMS program object, four ventilation regimes are created:

- 1. Outside of the working hours, when the office is not occupied, both the natural and mechanical ventilation are off, i.e. the windows are closed, and the HVAC system is not active.
- 2. During the working hours, when the thermal comfort of the occupants is not satisfactory, the HVAC system is turned on and the windows are closed. Thermal comfort is assessed using adaptive comfort model proposed in the ASHRAE-55 standard. (ASHRAE, 2017) The thermal comfort is considered unsatisfactory when less than 90% of the occupants is satisfied with the thermal environment (90% acceptability status).
- 3. During the working hours, when the occupants are satisfied with their thermal comfort (again, assessed using the 90% acceptability status) and the outdoor temperature is lower than the indoor one, natural ventilation is turned on, i.e. windows are opened. At the same time, HVAC system is turned off.
- 4. During the working hours, when the occupants are satisfied with their thermal comfort and the outdoor temperature is higher than the indoor one, both systems are turned off, i.e. the windows are closed, and the HVAC system is not active.

These four regimes were also based on the work by Neves et al. (2019). However, the authors use ASHRAE-55 adaptive model with 80% acceptability status. This way, by the time when HVAC system is activated, less than 80% of the people is satisfied with the thermal conditions. Therefore, this work uses 90% acceptability status as it gives higher satisfaction of occupants with the indoor environment, even at the expense of higher cooling loads. For the purpose of the thermal comfort assessment, it was assumed that the clothing of occupants is equal to 0.5 clo during the whole year.

At the end of the energy simulation, EnergyPlus reports the chosen variables, which include variables needed for later stages of model development: electricity used for lighting and equipment, and energy loads for cooling and heating.

## 3.4 Material modeling

The material data considered in this work is primarily based on commercial building types specified in the Brazilian national standard NBR 12721. (ABNT, 2005) This data source is convenient because it reflects typical national practices in the construction sector. It also groups the materials into categories that include similar items. (Sinduscon-MG, 2007) For simplification, NBR 12721 assumes that the building has a simple foundation type, and it excludes some equipment types (e.g. elevators, air-conditioning units, ventilation ducts).

The material data is listed in Appendix 1 in the form of material intensity values (weight of a given material per each  $m^2$  of floor space). The data is given for three building archetypes considered in this study, referred to as OPL-8, CL-8, and CL-16, which were created based on the NBR 12721 data (see 3.1.2).

The data was complimented by some industry data to add two more materials, common in office buildings: floor covering (carpet tiles) (Forbo, 2016) and acoustic ceiling (Knauf A/S, 2016). These two materials were additionally included because such layers decrease the extent to which the thermal mass of the concrete slab can be utilized, thus influencing energy performance of the building.

On top of the conventional materials, the data in Appendix 1 includes devices such as elevators and AC units, whose amount as measured in  $kg/m^2$  of floor space was added based on data from Sinduscon-MG (2007) and Ecoinvent database (Primas, 2010).

It is assumed that the material intensity data includes on-site material losses. In case of NBR 12721, this assumption can be supported by the fact that the dataset was created for project pricing calculations, and material losses make up a part of project costs.

The materials considered in this study can be divided into categories, according to the data processing procedure, as can be seen in Figure 3. The procedure is intended to create final material data from the initial material data (as listed in Appendix 1) and/or from material data exported from EnergyPlus. Material data extraction from EnergyPlus is done using BuildME framework developed by Heeren (2019) in order to calculate building material from EnergyPlus input files.



#### Figure 3 Material types according to the data processing procedure.

Type 1A materials, such as concrete and fiber cement, are partly included in EnergyPlus i.e. the material amount used in EnergyPlus is lower than the total amount of material as reported in the initial data. The difference is compensated for, so that the final material amount is equal the initial amount. In case of fiber cement roof tiles, this difference can be attributed to the fact that a flat roof with non-overlapping tiles was modeled, which may not reflect the real design of the buildings used to develop the initial amount. In case of concrete, the difference between the initial data and the material amount reported by EnergyPlus comes from the fact that only concrete floor slabs are modeled in EnergyPlus, while other parts of the concrete structure, such as load-bearing columns and foundations, were not modeled. The external floor slab was modeled two times thicker

than internal floor slabs to account for higher thermal mass associated with foundations. Such a simplification is typical for energy simulations and should not significantly influence the energy performance of the building. Both for concrete and fiber cement tiles, the difference was assumed to also include on-site material losses.

Type 1B materials are similar to type 1A materials, with the difference that EnergyPlus uses material data which is a mixture of a few materials in the initial data. This is the case for cement, hydrated lime and sand, whose mixture makes up plaster mortar, used on walls modeled in EnergyPlus. To divide plaster into these constituent materials, a ratio of 1:2:10 was used (cement to lime to sand), according to Brazilian guidelines for masonry workers. (Thomaz et al., 2009, p. 41) Again, as with the type 1A materials, the difference between EnegyPlus data and the initial material amount is compensated for, so that the final material amount is equal to the initial material amount.

Type 2 materials, such as brick, floor covering (carpet), and acoustic ceiling, are modeled in EnergyPlus in such a way that the material amount exported from EnergyPlus is exactly the same as the initial material amount. As a consequence, the data does not need to be additionally adjusted. This type of materials involves some simplification, which comes from the fact that initial material data is assumed to include also on-site material losses, while here, 100% of these materials is included in the building simulation. However, in practice, carpet and acoustic ceiling are typically not used everywhere in the building, which is not accounted for in the simulation. On top of that, mortar between bricks is also not modeled in EnergyPlus which treats the whole wall as being a homogenous brick wall. Such simplifications do not significantly influence the energy loads in the building, but they result in slightly overestimated amount of these materials. This difference could roughly account for the on-site material losses.

Type 3 materials are those for which no initial material data was used, which is the case of glass and aluminum, making up windows. The material intensity is calculated based on EnergyPlus data, and the result is dependent on a given window-to-wall ratio (WWR), as listed in Table 2. Assuming glass density of 2500 kg/m<sup>3</sup> and glazing thickness of 6 mm, there is 15 kg of glass for every 1 m<sup>2</sup> of glazing. Based on data for the commercial building archetypes of normal standard in NBR 12721, it is assumed that there is 4 kg of aluminum for each 1 kg of glass. (ABNT, 2005) These values are recalculated to kg/m<sup>2</sup> of floor space, so that the final data includes material intensity of glass and aluminum.

Finally, type 4 materials include ones that are not directly modeled in EnergyPlus as they do not have any meaningful influence on the energy loads of the system. These materials are generally used in smaller amounts so, even if they are applied to the building envelope, their contribution to the total thermal performance of the building envelope is negligible. All type 4 materials are directly added to the final material data. Examples of these materials include steel (acting as concrete reinforcement), doors, paint, and plastic pipes. Devices such as elevators and AC units are also treated as Type 4 materials.

The procedure of data processing described above and shown in Figure 2 allows to reflect the material data found in sources such as NBR 12721 (ABNT, 2005), while at the same time it ensures that the material-energy interaction in buildings is also taken into account. Such interaction is particularly evident in case of materials making up the building envelope and/or the thermal mass of the building. For example, the material intensity of glass and aluminum is directly related to the window-to-wall ratio of the building, and thus, influences its operational energy use. Another example could be bricks, whose amount used in external walls was smaller than the total brick material intensity, so internal walls were explicitly modeled (see 3.3) to meet the material intensity value, and thus account for additional thermal mass. Such an approach ensures that emission modeling is based on reliable data, in which the energy-material interaction is included.

### 3.5 Emission modeling

During the previous phases of model development, energy and material flows were quantified, allowing for GHG emission modeling. The emission model is built according to the European Standard EN 15978, which sets guidelines for assessment of environmental performance of buildings, based on life cycle assessment (LCA) methodology. (CEN, 2012) The standard specifies the approach that should be taken to quantify the flows of mass and energy associated with the building, all of which determine its environmental impact. It also describes the way in which the boundaries of the system should be set. The approach presented in EN 15978 is widely accepted in the scientific community, also within Brazil. (Gomes et al., 2018; Morales et al., 2019; Rohden and Garcez, 2018) Therefore, application of the same approach ensures comparability with the work of other researchers.

Life-cycle stages and particular modules of construction products, according to EN 15978 (CEN, 2012), can be found in Table 2. This work investigates the chosen stages: stage A (pre-use stage), stage B4 (replacement), stage B6 (operational energy use) and stage C (end-of-life stage). Modules B1, B2, B3, B5 and B7 were not included in the study. These modules are typically associated with low impacts, so they are rarely included in whole building LCA studies. (Frischknecht et al., 2019; The Carbon Leadership Forum, 2018) Stage D is related to reuse, recycling, and energy recovery outside the system boundary. As the potential for material and energy recovery from construction and demolition waste is very limited in the Brazilian reality (Gomes et al., 2018), stage D was also excluded from the study.

	A1	Raw material extraction and processing	
Product stage	A2	Transport to the manufacturer	
	A3	Manufacturing	
Construction process stage	A4	Transport to the building site	
construction process stage	A5	Installation into the building	
	B1	Use or application of the installed product	
	B2	Maintenance	
	B3	Repair	
Use stage	B4	Replacement	
	B5	Refurbishment	
	B6	Operational energy use	
	B7	Operational water use	
End-of-life stage	C1	Deconstruction, demolition	
	C2	Transport to waste processing	
	С3	Waste processing for reuse, recovery and/or recycling	
	C4	Disposal	
Benefits/loads beyond the system boundary	D	Reuse, recovery and/or recycling potentials	

Table 3 Life-cycle stages of construction products according to EN 15978. (CEN, 2012)Highlighted modules are the ones investigated in this study.

The system boundary for the lifecycle modeling is in line with the boundaries for each considered module, as defined by EN 15978 (CEN, 2012). The functional unit is defined as

 $1 \text{ m}^2$  gross floor area of a building during 1 year of operation. The considered environmental impacts are limited to climate change impacts measured using GWP100 metrics, as given by the ReCiPe midpoint method (H) V1.13. The life cycle inventory datasets used in the study were sourced from ecoinvent v3.6 database (allocation cutoff), complemented by some industry data in case a given process could not be found in ecoinvent. The life cycle inventory datasets can be found in Appendix 3, 4 and 5.

The total impact  $d_i$  for a given item *i* is generally calculated using the impact multiplier  $m_i$  and the external demand  $y_i$ :

$$d_i = GWP_i \cdot y_i \tag{1}$$

The elementary impact  $GWP_i$  is derived from life cycle inventory datasets and it stands for the GWP impact per unit of item *i*, measured in "kg CO<sub>2</sub>-eq/unit". The external demand  $y_i$ stands for the amount of item *i* associated with one functional unit, i.e. 1 m<sup>2</sup> gross floor area of a building during 1 year of operation. Thus, the external demand  $y_i$  is measured in unit/(m<sup>2</sup>.year). In some cases, the calculation of  $y_i$  requires an additional adjustment of the units. This is the case for materials or construction and demolition processes, which need to be divided by the total building lifetime.

The following chapters describe the assumptions made for the considered life cycle stages of the building.

#### 3.5.1 Modules A1-A4: Production and transport to the building site

Modules A1-A4 are associated with the production and transport of materials used in the construction of buildings. The impact associated with the production of material *i* can be calculated using equation (1). The elementary impact  $GWP_i$  is derived from the production process dataset for material *i* found in Appendix 3. The external demand  $y_i$  is the final material amount calculated in section 3.4 divided by the building lifetime, in order to adjust the units in equation (1). As explained during material modeling description in section 3.4, for most materials (except for glass and aluminum), the final material amount is equal to the initial material amount as listed in Appendix 1.

The life cycle inventory datasets for material production processes were chosen to best reflect the original data from NBR 12721 (ABNT, 2005). For example, as described in NBR 12721 and the complementary report from Sinduscon-MG (2007), aluminum was in the form of window frames, so the production process chosen for this material was *market for window frame, aluminium*, U=1.6 W/m2K, *GLO*. When no material details were available in NBR 12721 or the data was non-conclusive, country-specific reports were used for reference. For example, concrete of compressive strength of 20 MPa was reported in NBR 12721 (ABNT, 2005), while concrete of 25 MPa was reported by Sinduscon-MG (2007). Based on national data on concrete consumption, concrete of 25 MPa of compressive strength was chosen, as it is the most common in Brazil. (Silva et al., 2018, Tab. 14)

During dataset selection, country-specific datasets were preferred. If no Brazilian process was available, other datasets from ecoinvent database were chosen. In case of acoustic ceiling, no appropriate ecoinvent process was found, and no Brazilian industry data was available. In this situation, industry data from Denmark was used, in form of an environmental product declaration (EPD). (Knauf A/S, 2016)

Ecoinvent database v3.6 offers a process type known as *market for product*, which includes a typical consumption mix of a given product, together with average transport values and losses in trade and transport. Depending on data availability, these market processes may

relate to the global market or a given country's market. (Ecoinvent, n.d.) All the system processes selected for this study are of the *market* type, and so they include unit processes related with the transportation to the construction site. Some of the system processes are Brazilian, reflecting average transportation distances for a given type of product. Others are global processes, so the transportation distances may be not in line with Brazilian reality.

The ready-made datasets used for life cycle inventory included 'cradle to gate' unit processes for material production, as defined by EN 15978 (CEN, 2012). Additionally, the choice of *market* process type assures that the transport to the building site is also included in the dataset. As such, the datasets cover life cycle modules A1-A4.

#### 3.5.2 Module A5: Installation into the building

According to EN 15978, module A5 includes multiple processes related to the construction process, e.g. ground works, on site product transformation, installation of products into the building, waste management of construction waste. (CEN, 2012) It was assumed that most of the climate change impact originates from construction machinery operating on site and the energy use associated with it. Therefore, waste management of construction waste (related to on-site material losses) was not included.

The original data from the Brazilian national standard NBR 12721, used for office building archetype definition, lists the operation time of construction machinery, taking a concrete mixer of 320 liters as an approximation for all the machinery. (ABNT, 2005) According to market data, such mixers are usually powered by electricity and have around 2 hp (horsepower) of power. Using this information, energy used in construction of 1 m<sup>2</sup> of floor space was calculated for each building archetype, as seen in Table 4.

Concrete mixer, 320 I	OPL-8	CL-8	CL-16
Operation time (days/m <sup>2</sup> ) (ABNT, 2005)	0.64	0.27	0.64
Operation time (hours/m <sup>2</sup> )	15.32	6.52	15.32
Energy used (kWh/m <sup>2</sup> )	22.86	9.73	22.86

#### Table 4 Energy use for construction of office building archetypes.

The calculated energy use for construction processes associated with a chosen building archetype is the external demand  $y_i$  in equation (1), which can be used to calculate climate change impacts. The elementary impact  $GWP_i$  is derived from the electricity process datasets found in Appendix 5. The processes are region-dependent, so the elementary impact of electricity depends on the choice of a city, as listed Table 2. The external demand  $y_i$  needs to be divided by the building lifetime, in order to adjust the units in equation (1).

Electricity used by the machinery is complemented by impacts associated with the production of machinery itself. The impacts were calculated from equation (1), using an ecoinvent dataset for building machinery (Appendix 4). It was assumed that 1.34E-07 unit of a building machine is used for each 1 MJ of energy used. (Kellenberger, 2010) Again, the external demand  $y_i$  needs to be corrected for the building lifetime.

#### 3.5.3 Module B4: Replacement

Replacement module includes the production of the replaced product and its transportation to the site, as well as waste management of the removed product. (CEN, 2012)

The impact  $d_i$  of replaced material *i* is calculated using a modified version of equation (1):

$$d_i = GWP_i \cdot y_i \cdot \left(\frac{BL}{ML_i} - 1\right) \tag{2}$$

The elementary impact  $GWP_i$  is derived from *both* the production and end-of-life process dataset for material *i* found in Appendix 3. The total elementary impact  $GWP_i$  should be equal to the sum of production and end-of-life elementary impacts (as measured in kg  $CO_2$ -eq/kg of item *i*). Analogically to calculation for modules A1-A4 described in section 3.5.1, the external demand  $y_i$  is the final material amount calculated in section 3.4 divided by the building lifetime. The last factor in equation (2) stands for the number of replacements during the building lifetime BL, determined by the material lifetime ML. Each material *i* has its own service life  $ML_i$ , listed in Appendix 2. As can be evident from the equation, materials whose lifetime is equal to the building lifetime are not being replaced.

The material lifetime multiplier is applied at this point. It is a variable explained in section 3.1.6, whose values are listed in Table 2. The material lifetime  $ML_i$  is multiplied by the material lifetime multiplier (75%, 100% or 125%) which results in faster, typical, or slower replacement cycles, respectively. As already mentioned in section 3.1.6, materials whose lifetime is equal to the building lifetime are not subject to change.

It was assumed that replacement can take on fractional values. Although this may seem counterintuitive, this approach is logical considering the full picture. Assuming that the expected building lifetime will have been over in 10 years, how many facility managers would decide to replace an elevator, whose lifetime is 25 years? In such a case, a fractional replacement value calculated for the remaining lifetime (10/25 = 0.4) would be equivalent to 40% of facility managers deciding to replace the elevator.

A particular situation could also take place: if a building lifetime of 50 years is adopted while the material lifetime multiplier is 125%, some materials could have higher lifetime than the building itself (e.g. aluminum). Nonetheless, it was decided that the full burden of this material should be placed upon the building, as it is unlikely that such a product would be used in another building for the remaining lifetime.

#### 3.5.4 Module B6: Operational energy use

The environmental impacts related to operational energy use were calculated using data obtained during energy modeling described in section 3.3. EnergyPlus simulation results include the electricity used for lighting and equipment, and energy loads for cooling and heating. The heating and cooling loads are translated to electricity use using efficiency values described in section 3.1.4: heating efficiency of 100% and a chosen cooling efficiency value in range of 2 to 4 W/W (cooling efficiency is a variable parameter, as seen in Table 2).

The four components of electricity consumption (lighting, equipment, cooling, heating) are measured in kWh/m<sup>2</sup>/year. Using these electricity end-uses as external demand values  $y_i$ , the calculation of impact from equation (1) is straightforward. The elementary impact  $GWP_i$  is derived from the electricity process datasets found in Appendix 5. The processes are region-dependent, so the elementary impact depends on the choice of the city, which is a variable parameter as shown in Table 2.

#### 3.5.5 Module C1: Deconstruction, demolition

Module C1 includes all on-site operations which are necessary for the deconstruction and/or demolition processes. (CEN, 2012) The initial dataset from NBR 12721, used for office building archetype definition, does not include any information on demolition processes. However, a guide called TCPO (Price Composition Tables for Budgets) presents Brazilian averages of material and equipment needs for construction and demolition of buildings. (PINI, 2010)

Concrete makes up the great majority of total material weight in the office building archetypes (Appendix 1), so it is used as an approximation of demolition processes. Other building elements of Brazilian building (e.g. brick walls) are often dismantled manually (PINI, 2010, pp. 58–62), often due to lower costs of manual deconstruction. (da Rocha and Sattler, 2009) According to the TCPO guide, the machinery used for reinforced concrete demolition is a hydraulic breaker with an air compressor of 47 kW, powered by diesel. The air compressor is used for 5 hours to demolish 1 m<sup>3</sup> of reinforced concrete. (PINI, 2010, p. 59) Assuming concrete density of 2380 kg/m<sup>3</sup>, the energy used for demolition can be calculated as seen in Table 5.

Air compressor, 47 kW	OPL-8	CL-8	CL-16
Operation time (hours/m <sup>3</sup> concrete) (PINI, 2010)	5.00	5.00	5.00
Concrete material intensity (kg/m <sup>2</sup> ) (ABNT, 2005)	895.40	885.05	1247.02
Operation time (hours/m <sup>2</sup> )	1.88	1.86	2.62
Energy used (MJ/m <sup>2</sup> )	318.28	314.60	443.27

Table 5 Energy u	se for demolition	of office	building	archetypes.
------------------	-------------------	-----------	----------	-------------

The calculated energy use for demolition processes associated with a chosen building archetype is the external demand  $y_i$  in equation (1), which can be used to calculate climate change impacts. The elementary impact  $GWP_i$  for diesel is derived from the process dataset found in Appendix 4. The external demand  $y_i$  needs to be divided by the building lifetime, in order to adjust the units in equation (1).

The diesel dataset already includes machinery wear due to operation, which could be an approximation for wear of the referenced hydraulic breaker with an air compressor.

#### 3.5.6 Modules C2-C4: Waste processing and disposal

The last building life cycle stage to be considered is related to waste transportation from the building site, waste treatment, processing, and disposal. (CEN, 2012) The procedure is analogical to the one for modules A1-A4 described in section 3.5.1, with two main differences: (1) instead of production processes, disposal processes were considered; (2) transportation impacts were calculated manually.

The impact associated with the disposal of material *i* can be calculated from equation (1). The elementary impact  $GWP_i$  is derived from the disposal process dataset for material *i* found in Appendix 3. The external demand  $y_i$  is the final material amount calculated in section 3.4 divided by the building lifetime, in order to adjust the units in equation (1).

Brazilian datasets were chosen for the disposal processes whenever possible. According to process data of one of these Brazilian datasets, the typical disposal mix is composed of: 88% unsanitary landfill, 8% open dump, 3% open burning, and 1% municipal incineration.

(Symeonidis, 2019) Therefore, the waste products for which no Brazilian dataset could be found were modeled in a way to resemble this Brazilian reality to a greatest extent possible. Unfortunately, unsanitary landfill processes were available only for materials with readymade Brazilian disposal datasets. In this case, inert material landfill was chosen where applicable. (Doka, 2009, p. 27) If a given product was made of multiple materials, the material making up the majority of the product was chosen for the disposal process. For example, a waste gypsum process was chosen for gypsum plasterboard and a waste wood process was chosen for wooden doors.

In modules A1-A4 described in section 3.5.1, market processes were chosen to include transportation distances in the ready-made datasets. However, a majority of materials does not have country-specific datasets, and adoption of global disposal datasets would mean that the disposal methods are far from Brazilian reality. Additionally, the Brazilian disposal datasets include an industry average of 77 km of transportation distance. (Symeonidis, 2019) However, most of the considered office buildings would likely be located in big cities, where disposal sites are nearby. For example, Condeixa et al. (2014) assume 11 km of distance to disposal site for a building located in Rio de Janeiro.

Due to the above reasons, transportation was considered separately. For non-market processes, transportation of 11 km was added for each kg of material. For market processes, the distance was corrected to 11 km. The mode of transport was adopted after (Symeonidis, 2019), who used market for transport, freight, lorry, unspecified, RoW to model the transportation of demolition waste. The same process was adopted in this study, as road freight is particularly common in Brazil. (Condeixa et al., 2014; Gomes et al., 2018)

#### 3.5.7 Emission modeling summary

The previous subsections presented the process of GHG emission modeling. The modeling was performed using data on flows of energy and materials, quantified at the previous stages of model development, as was shown on the diagram in Figure 2. The operational energy, whose modeling was described in section 3.3, was used to calculate the impacts associated with module B6. On the other hand, the materials whose modeling was described in section 3.4 were used to calculate the impacts related to modules A1-A4, B4 and C2-C4. The impacts of modules A5 and C1 were calculated independently.

The individual impacts of modules described above can be summed up to obtain the total climate change impacts, also known as the carbon footprint. The total impacts can be calculated for all the buildings selected in the sampling process. The next step is then to analyze the results using methods of sensitivity analysis.

#### 3.6 Sensitivity analysis

The life cycle assessment done in the previous stage results in a whole range of climate change impact values for the buildings. Understanding the significance of the variable parameters on the GWP impacts requires a statistical analysis of the results.

A multivariate regression analysis was used to quantify the sensitivity of the results to the input variable parameters. The regression was based on a linear regression equation, adopted from Hygh et al. (2012):

$$y(x_1, x_2, \dots, x_{10}) = \beta_0 + \sum_{i=1}^{10} \beta_i x_i$$
(3)

where y is the dependent variable, such as the total GWP impact;  $x_i$  represents the value of a variable parameter *i* and  $\beta_i$  is the corresponding regression coefficient. The coefficient of determination (denoted R<sup>2</sup>) describes goodness-of-fit i.e. how close the datapoints are to the fitted regression line. Adjusted R<sup>2</sup> is often used instead of R<sup>2</sup> to account for the number of independent variables in the model.

The sensitivity analysis results are interpreted using standardized regression coefficients. There are two ways of obtaining these standardized coefficients, with both yielding the same results: 1) standardizing all the variables prior to the regression analysis; 2) performing the regression analysis on unstandardized variables and then adjusting the coefficients using standard deviation values of the variables. (Bring, 1994) In this work, the first method is used, even though Hygh et al. (2012) used the second method. As such, the standardized regression coefficients were calculated from formulas adopted from Bring (1994):

$$x_i^* = \frac{x_i - \bar{x}_i}{x_i} \tag{4}$$

$$y^* = \frac{y - \bar{y}}{s_y} \tag{5}$$

where  $x_i^*$  and  $y^*$  are standardized variables,  $\bar{x}_i$  and  $\bar{y}$  are the means of each variable in the sample and  $x_i$  and  $s_y$  are the standard deviations.

Therefore, the final form of the regression equation is:

$$y^*(x_1, x_2, \dots, x_{10}) = \sum_{i=1}^{10} B_i x_i^*$$
(6)

where  $B_i$  are the standardized regression coefficients for a variable parameter *i*. Note that the intercept coefficient  $\beta_0$  found in equation (3) vanishes due to standardization.

The standardized coefficients obtained in this way can quantify parameter sensitivity by providing information on the relative impact of each parameter. These coefficients are dimensionless which allows for direct comparison between the variables, even if they have different units. The same would not true for unstandardized coefficients. The standardized coefficients reflect the sensitivity of the dependent variable y (e.g. the total GWP impact) on changes in values  $x_i$  of the variable parameters i. The bigger the absolute value of coefficient  $B_i$ , the more sensitive y is to the changes in parameter i. The sign of  $B_i$  indicates the correlation i.e. if the coefficient  $B_i$  is positive, then an increase in parameter i will cause the dependent variable y to increase as well.

To perform a regression analysis, all the categorical variables need to be described using numeric data. In case of this study, there are two categorical parameters: city and building type, as seen in Table 2.

Climate was simulated by choosing one of 12 Brazilian cities. However, as it turned out during emission modeling, electricity mix is dependent on the region (see 3.5.4). As a result, the choice of a particular city has two different effects on the model: 1) it influences the climatic conditions which determine the energy need of the building; 2) it influences the electricity mix which determines the emissions associated with each kWh of electricity used. Therefore, it was decided that these two effects should be separated for a better understanding of the model's results. During the regression analysis, they are treated as independent variables.

Climatic conditions are characterized numerically using annual cooling degree hours (CDH) from Versage et al. (n.d.), who calculated CDH with a wet-bulb temperature of 15°C. Such metric was shown by Borgstein and Lamberts (2014) to be the most appropriate for describing the cooling energy demand of Brazilian climates. Electricity mix is characterized by elementary GWP impact of 1 kWh of electricity. The values of CDH and elementary GWP impacts of electricity for each investigated city is listed in Appendix 5.

Building type was changed to numeric data by creating two new 'dummy' variables with a binary notation: CL-8 and CL-16. Each of these binary variables will have a value of 1 if a given sample item is of the given building type and a value of 0 otherwise. As a consequence, if for a given sample item both CL-8 and CL-16 variables are zero, then this sample item belongs to the third building type (OPL-8). This way, the standardized regression coefficients for variables CL-8 and CL-16 should reflect the change in the dependent variable y associated with switching the building type from OPL-8 to CL-8 or CL-16.

The final list of 12 independent variables used during the regression analysis can be found in Table 6.

Independent variable	Values		
Electricity mix	According to the city chosen from Table 2,		
Climate (CDH)	values listed in Appendix 5		
CL-8	$P_{inary}(0, ar, 1)$ if both are 0 then building type is $OP_{inary}(0, ar, 1)$		
CL-16	Binary (0 of 1), it both are 0 then building type is OPE-8		
WWR	30%, 50%, 70%, 90%		
SHGC	0.2, 0.4, 0.6, 0.8		
Window opening effective area	0.0, 0.1, 0.3, 0.5, 0.7, 0.9		
Shading	0.0, 0.2, 0.4, 0.6, 0.8, 1.0		
Cooling setpoint	22, 23, 24, 25		
Cooling efficiency	2.0, 2.5, 3.0, 3.5, 4.0		
Building lifetime	50, 75, 100		
Material lifetime multiplier	75%, 100%, 125%		

Table 6 Inde	pendent variables	used in the re	egression analy	vsis. Derived from <sup>•</sup>	Table 2.
Table 0 Inde	pendent variables	used in the re	sgression analy		

The sensitivity analysis procedure described above gives a quantitative measure of parameter sensitivity. Variable parameters analyzed in this study can be ranked according to their influence on the results, allowing for interpretation of the results obtained in the previous stages of model development.

# 4 Results

The methodology described in the previous chapter served as a means of exploring Brazilian office buildings and their carbon footprint. The office building archetypes allowed for a selection of 1000 buildings which were analyzed in terms of energy and materials, eventually allowing for climate change impact assessment. This chapter presents results of these 1000 simulations in terms of energy, materials, and emissions. The plots are were generated using matplotlib and seaborn libraries for Python. (Hunter, 2007; Waskom et al., 2020) The results are also analyzed by means of a multivariate regression analysis.

### 4.1 Energy modeling results

The results of energy simulations of the 1000 sampled buildings were analyzed from the perspective of energy load – the amount of energy needed by the building to provide services to its occupants. Energy loads for cooling and heating do not include the efficiency of the HVAC system.

The total energy load in the simulated buildings is composed of equipment, lighting, cooling, and heating, as seen in Figure 4. It is characterized by high variability, with values ranging between 90 and 407 kWh/m<sup>2</sup>/year and a median of 206 kWh/m<sup>2</sup>/year. Converting the energy loads to delivered energy yields values between 82 and 238 kWh/m<sup>2</sup>/year.

The predominant end use is cooling with values from 10 to 331 kWh/m<sup>2</sup>/year, and a median of 126 kWh/m<sup>2</sup>/year. The energy need for equipment and lighting is constant, equal to 43.68 and 32.76 kWh/m<sup>2</sup>/year, respectively. The heating load is close to zero, although the exact value is also characterized by some variability, reaching up to 7 kWh/m<sup>2</sup>/year.

The cooling load as a function of city is shown in Figure 5. The cities are arranged from left to right in order of increasing CDH. Generally, the total energy load shows an increasing trend with increasing CDH. The median energy load values range from 113 kWh/m<sup>2</sup>/year in Curitiba to 290 kWh/m<sup>2</sup>/year in Belém.

However, city was just one of the variable parameters considered in this study, whose full list can be found in Table 2. Figure 6 shows the total energy load as a function of all the other variable parameters. For comparability, the y-axis is shared among all the parameters. However, the three bottom parameters (cooling efficiency, building lifetime, and material lifetime multiplier) should do not have influence on energy loads, as they were applied at later stages of the model development. The figure shows all the sampled buildings, which includes both mixed-mode and fully AC buildings.

The biggest influence on energy load can be observed for the window opening effective area, which reflects the effectiveness of natural ventilation strategies. The mean of energy load values ranges from 173 to 269 kWh/m<sup>2</sup>/year, with the lowest value for the highest window opening effective area, which was equal to 0.9. As the value of 0 means that windows cannot be opened, this reflects a case where only the HVAC system is active at all times and natural ventilation cannot be used. This is also the reason for the abrupt change between 0 and 0.1 for window opening effective area. A clear trend can also be observed for SHGC, shading, and cooling setpoint.



Figure 4 Total energy load and its components: equipment, lighting, cooling, and heating.



Figure 5 Total energy load as a function of city. The cities are arranged according to increasing cooling degree hours. (Versage et al., n.d.)



Figure 6 Total energy load as a function of the chosen variable parameters. Parameter 'city' is excluded as it is shown in another plot.

## 4.2 Material modeling results

Material modeling was mostly based on reconciling values from the initial datasets with the values extracted from EnergyPlus, in order to properly reflect energy-material interactions. As such, the material amounts follow the values from the original material dataset, as presented in Appendix 1. The exception is glass and aluminum, whose amounts are based on data extracted from EnergyPlus using BuildME framework developed by Heeren (2019).

Glass and aluminum are strictly related to the size of windows of the simulated building, and thus their amount is related to WWR, one of the variable parameters. The values for WWR and material intensity of glass and aluminum can be found in Table 7.

The amount of glass and aluminum increases with increasing WWR in a linear manner, with an increase of 1.40 kg/m<sup>2</sup> and 5.60 kg/m<sup>2</sup> for every 20% increase of WWR for glass and aluminum, respectively.

Window-to-wall ratio (WWR)	Glass (kg/m²)	Aluminum (kg/m²)
30%	2.10	8.40
50%	3.50	14.00
70%	4.90	19.60
90%	6.30	25.20

#### Table 7 Window-to-wall ratio and material intensity of glass and aluminum.

## 4.3 Emission modeling results

Based on the operational energy and the material amount calculated in previous steps, climate change impacts of the buildings were calculated, using GWP100 as metrics. The impacts according to life cycle stages divided by modules can be seen in Figure 7.

The total GWP values vary from 20 to 108 kg  $CO_2$ -eq/m<sup>2</sup>/year, with 41 kg  $CO_2$ -eq/m<sup>2</sup>/year median. Module B6 (operational energy use) is visibly the most impactful, characterized by a range of values from 11 to 95  $CO_2$ -eq/m<sup>2</sup>/year, with a median of 25  $CO_2$ -eq/m<sup>2</sup>/year. Module A1-A4 and B4, related with material production and material replacement, have smaller impacts, with medians of 8  $CO_2$ -eq/m<sup>2</sup>/year and 6  $CO_2$ -eq/m<sup>2</sup>/year, respectively. Modules A5, C1 and C2-C4 have negligible impacts, with medians of 0.01, 0.40 and 0.23  $CO_2$ -eq/m<sup>2</sup>/year, respectively. These modules refer to construction activities, demolition activities and end-of-life impacts associated with materials.

Module B6 concerns operational energy use, and so the emissions are largely dependent on the energy load values, as was shown in Figure 4. The exact relationship between the two variables is demonstrated in Figure 8. As a general rule, higher energy load values seem to be associated with higher GWP impacts, but some datapoints in the upper right of the plot are visibly distinct from this general trend. Consequently, as the energy loads increase, the possible range of GWP values for buildings of similar energy loads increases as well.

Modules A1-A4, associated with material production and transport to the building site, can be decomposed down to material level to investigate which materials have the biggest impact. The GWP impact of these modules is often referred to as embodied emissions. Figure 9 shows embodied emissions as a function of material, divided by building type. For clarity reasons, the selected plot type displays the mean and an indication of uncertainty but not the actual spread of the values.

Aluminum is responsible for the biggest share of embodied emissions, followed by steel, concrete, lime, and brick. The mean aluminum-related impact is 3 kg  $CO_2$ -eq/m<sup>2</sup>/year. There is also a visible difference between concrete and steel GWP impacts of 8-floor and 16-floor archetypes.

Module B4 is related to material replacement impacts. Its GWP impact is often referred to as recurrent emissions. Similarly to modules A1-A4, GWP impacts as a function of material and building type can be plotted, which was done in Figure 10. For comparison purposes, Figure 9 and Figure 10 have the same y-axis limits.

On average, aluminum and paint have similar recurrent emissions, with the mean value of around 2 kg  $CO_2$ -eq/m<sup>2</sup>/year. Following are the impacts associated with the AC system, floor covering, doors, and ceramic tile. A significant difference between the building types can only be seen for paint: the 16-floor archetype with a cellular layout (CL-16) has higher paint-related recurrent emissions than the other two archetypes.

The five most impactful materials from Figure 9 and Figure 10 are shown again in Figure 11 and Figure 12, respectively. This time, the plot type depicts more statistical details of the data. The values are divided by building lifetime.

Both Figure 11 and Figure 12 show high variability of aluminum-related impacts. Module A1-A4 has aluminum impacts in range of 1.0 to 6.4 kg  $CO_2$ -eq/m<sup>2</sup>/year, while module B4

in range of 0.0 to 6.4 kg  $CO_2$ -eq/m<sup>2</sup>/year. Other materials display much lower variations, usually enclosed within 1 kg  $CO_2$ -eq/m<sup>2</sup>/year of differences at most. Building lifetime has a clear impact on the GWP values: a higher lifetime decreases the embodied emissions but increases the recurrent ones.

Finally, the relationship between the variable parameters specified in Table 2 and GWP impacts can be explored. Figure 13 shows the total GWP as a function of city, represented by different cities. The cities are arranged according to their annual cooling degree hours, listed in Appendix 5. There is no visible trend between GWP and CDH, which contrasts the trend that was evident for energy loads in Figure 5. Three cities are particularly distinct from the others: Recife, Salvador, and Fortaleza, with median GWP values of around 70 kg  $CO_2$ -eq/m<sup>2</sup>/year. These cities are characterized by relatively high CDH values. However, there are two cities with even higher CDH values which nonetheless have much lower emissions (around 45 kg  $CO_2$ -eq/m<sup>2</sup>/year).

Figure 14 shows the total GWP impacts as a function of the variable parameters considered in this study, excluding city parameter which was already analyzed above. For comparability, the y-axis is shared among all the parameters. An increasing or decreasing trend can be seen for all the parameters, the trend is stronger for some and weaker for others.

The highest spread of possible emissions is observed for window opening effective area and cooling efficiency, with the mean values dropping by 16% from around 51 to around 43 kg  $CO_2$ -eq/m<sup>2</sup>/year as the parameter values increase. General decreasing trends can also be seen for shading, building lifetime, cooling setpoint and material lifetime multiplier. On the other hand, WWR and SHGC are generally associated with increasing emissions with increasing parameter values. CL-16, the only 16-floor archetype, shows higher GWP impacts than the other two archetypes.



Figure 7 GWP impacts of the building life cycle modules.



Figure 8 Relationship between the total GWP and the total energy load.



Figure 9 GWP of module A1-A4 as a function of material and building type.



Figure 10 GWP of module B4 as a function of material and building type.



Figure 11 GWP of module A1-A4 as a function of selected materials and building lifetime.



Figure 12 GWP of module B4 as a function of selected materials and building lifetime.



Figure 13 Total GWP impacts as a function of city. The cities are arranged according to increasing annual cooling degree hours. (Versage et al., n.d.)



Figure 14 Total GWP impacts as a function of the chosen variable parameters. Parameter 'city' is excluded as it is shown in another plot.

## 4.4 Regression analysis

The relationship between the variable parameters and the results can also be investigated by means of a multivariate regression analysis.

The results of a regression model for total energy loads can be seen in Table 8. CDH shows the highest influence on the energy loads, followed by window opening effective area, cooling setpoint, SHGC and shading. Similar results can be observed both for mixed-mode and for fully air-conditioned buildings, except for the fact that in fully AC buildings, window opening effective area is a constant, so it was excluded from the analysis. Small differences between the two ventilation modes can be observed for WWR and CDH – mixed-mode buildings are slightly less sensitive to these parameters than fully AC buildings.

The results of a regression model for GWP impacts can be seen in Table 9. The sensitivity analysis was performed for the total GWP impacts and for each of the modules that was shown to have non-negligible impacts according to Figure 7. Modules A1-A4 are the most sensitive to building lifetime, followed by WWR and CL-16. Module B4 is the most impacted by material lifetime multiplier, followed by building lifetime, WWR and CL-16. Module B6 and total GWP show very similar trends: they are strongly influenced by the electricity mix, followed by CDH, cooling efficiency, and window opening effective area. However, the total GWP shows more sensitivity than B6 some of the remaining parameters: WWR, material lifetime multiplier, building lifetime, and CL-16.

	Standardized coefficients				
Independent variable	All buildings	MM buildings only	AC buildings only		
Electricity mix	-0.03	-0.04	0.02		
Climate (CDH)	0.82	0.88	0.91		
CL-8	-0.01	-0.02	0.04		
CL-16	-0.01	-0.02	0.02		
WWR	0.08	0.07	0.12		
SHGC	0.14	0.15	0.16		
Window opening effective area	-0.42	-0.31	-		
Shading	-0.09	-0.11	-0.10		
Cooling setpoint	-0.19	-0.20	-0.20		
Cooling efficiency	0.00	-0.01	0.00		
Building lifetime	0.00	-0.01	0.00		
Material lifetime multiplier	-0.01	-0.01	0.00		
	Adj. R <sup>2</sup> = 0.934	Adj. R <sup>2</sup> = 0.945	Adj. R <sup>2</sup> = 0.977		

#### Table 8 Results of the multivariate regression analysis of total energy loads.

Tudayandantan sista	Standardized coefficients				
Independent variable	A1-A4	B4	B6	Total GWP	
Electricity mix	0.00	0.01	0.80	0.79	
Climate (CDH)	0.01	-0.02	0.21	0.21	
CL-8	0.02	0.01	0.00	0.00	
CL-16	0.30	0.21	-0.01	0.08	
WWR	0.42	0.28	0.02	0.14	
SHGC	0.00	0.00	0.06	0.06	
Window opening effective area	0.00	0.00	-0.14	-0.14	
Shading	-0.01	0.01	-0.03	-0.04	
Cooling setpoint	0.00	0.00	-0.07	-0.07	
Cooling efficiency	-0.01	0.01	-0.18	-0.17	
Building lifetime	-0.84	0.41	0.01	-0.08	
Material lifetime multiplier	0.00	-0.82	0.00	-0.12	
	Adj. R <sup>2</sup> = 0.952	Adj. R <sup>2</sup> = 0.938	Adj. R <sup>2</sup> = 0.957	Adj. R <sup>2</sup> = 0.957	

Table 9 Results of the multivariate regression analysis of GWP impacts.

# 5 Discussion

This chapter discusses how the results presented in the previous chapter allow to answer the research questions. Further, limitations of the study are listed and suggestions for future work are given.

## 5.1 Range of carbon footprint values

The first question of the study sought to determine the possible range of carbon footprint values for Brazilian office buildings. The results of the study show that the total GWP values vary from 20 to 108 kg  $CO_2$ -eq/m<sup>2</sup>/year, with 41 kg  $CO_2$ -eq/m<sup>2</sup>/year median. These values cannot be compared to any other Brazilian work on office buildings, as there was no study that would show the impacts of the whole life cycle of an office building. Foreign studies were not used because the GWP impact is dependent on climate and other regional factors. The values found in Brazilian literature for other types of buildings show that they are generally in the same order of magnitude. (Evangelista et al., 2018; Gomes et al., 2018)

The distribution of the total GWP impacts across different life cycle stages and modules was shown in Figure 7. The importance of different modules is discussed in the following paragraphs, in the order of decreasing significance for the total GWP values, as observed in Figure 7: operational energy use (B6), material production and transport to the building site (A1-A4), replacement (B4) and the remaining modules (A5, C1, C2-C4).

*Operational energy use (B6)* was the biggest contributor to the total GWP impact, and at the same time the main source of impact variations. This is an expected outcome, as it widely known in the research community that operational energy is the biggest contributor to environmental impacts of typical buildings. (Ramesh et al., 2010) The whole range of delivered energy values was found to cover values typically found in the Brazilian literature (Alves et al., 2017; Carvalho et al., 2010; Wong et al., 2019), with the exception of some bank branches (Borgstein and Lamberts, 2014) and buildings with special equipment such as data centers (Lamberts et al., 2015). This is a result of the fact that the model included only typical office equipment. Unfortunately, the emissions associated with this stage could not be compared with any literature values for similar reasons as explained above for the total GWP impact. The Brazilian studies that were used there (Evangelista et al., 2018; Gomes et al., 2018) cannot be used for module B6 due to strongly different assumptions on energy use.

The operational energy use was found to be strongly dependent on cooling energy load, which was subject to significant variations (Figure 4). This variability can be mostly attributed to different climatic conditions of the chosen cities (Figure 5), which is additionally supported by the regression analysis (Table 8). Comparison of the findings with those of Borgstein and Lamberts (2014) confirms that annual cooling degree hours measured using a wet-bulb temperature of 15°C can be a good predictor of energy loads in a given climate.

Among variable parameters other than the city, window opening effective area had the strongest effect on annual energy loads (Figure 6), which confirms the findings of Neves et al. (2019). The relation between the total energy load and SHGC, shading, WWR, and

cooling setpoints reflects the trends observed by other researchers. (Neves et al., 2019; Santesso and Chvatal, 2018)

The results of the sensitivity analysis (Table 8) enable to conclude that energy loads in mixed-mode and fully AC buildings are sensitive to practically the same similar parameters. The only difference is window opening effective area (constant for fully AC buildings). The regression model also shows that energy loads were almost independent of the building type, suggesting that differences between the archetypes (a greater number of floors in CL-16, less thermal mass in OPL-8) did not significantly affect their energy performance.

*Material production and transport to the building site (A1-A4)* was the second most important factor in the total GWP impact, according to median values (Figure 7). The median impact was around 8 CO<sub>2</sub>-eq/m<sup>2</sup>/year. This value is of the same order of magnitude as values found in other studies on Brazilian residential buildings. (Evangelista et al., 2018; Morales et al., 2019) This finding is also in line with embodied GWP impacts of commercial buildings available in deQo database which lists 160 commercial buildings in 10 countries. (De Wolf, n.d.) On the other hand, even considering the full variance, the obtained values were smaller than the impact found by Gomes et al. (2018) in their study on a university building in Brazil. However, this discrepancy is easily explained by the fact that Gomes et al. (2018) studied a net zero energy building, characterized by much higher embodied impacts due to special energy-efficient design and photovoltaic panels.

The impact of modules A1-A4 was found to be dominated by emissions associated with aluminum, followed by steel, concrete, lime and brick (Figure 9). It is somewhat surprising that aluminum is the most impactful material, because various LCA studies have shown that concrete and steel are the two biggest contributors to climate change impacts. (Evangelista et al., 2018; Morales et al., 2019; Saade et al., 2014; Silva, 2013) However, none of these studies was specific to office buildings. On the other hand, the work by Najjar et al. (2019) was focused on Brazilian office building structures and it found that the "modern" building alternative (with a curtain wall system) was associated with extremely high aluminum emissions. Studies on façade solutions have also shown a particularly high levels of emissions for aluminum. (Brugnera, 2018; Taborianski and Prado, 2012)

Archetype CL-16 showed higher embodied emissions than OPL-8 and CL-8 for materials such as concrete, steel, and bricks (Figure 9). This is a direct result of higher material intensity values for CL-16 compared to OPL-8 and CL-8, which is associated with its height (16 floors). The fact that material intensity for structural materials increases with the building height has been mentioned by various researchers. (De Wolf, 2014; Foraboschi et al., 2014; Treloar et al., 2001) The same studies also show that taller buildings are associated with higher embodied energy. Therefore, a logical consequence of this fact is that taller buildings have higher embodied emissions, which was observed in the current study. Results of the regression model also confirm that archetype CL-16 has more impacts for modules A1-A4 (Table 9).

As expected, embodied emissions displayed a clear relationship with building lifetime: the longer the lifetime, the lower the embodied emissions (Figure 11). The widest spread of possible embodied emissions was observed for aluminum, which can be directly attributed to the fact that aluminum was modeled a linear function of WWR (Table 7).

*Replacement (B4)* was the next module with a relatively high contribution to the total GWP impact. The median impact was around 6  $CO_2$ -eq/m<sup>2</sup>/year. This value is of the same order

of magnitude as replacement emissions found by Morales et al. (2019) for social housing in Brazil.

These material replacement emissions, also referred to as recurrent embodied emissions, were dominated by aluminum and paint, followed by the AC devices, floor covering, doors, and ceramic tile (Figure 10). Except for aluminum, these are generally materials with minor importance for initial embodied emissions, whose high importance for recurrent emissions results strictly from relatively frequent replacements. The most striking difference between initial and recurrent embodied emissions appears to be the one for paint whose median impact increases by almost 9 times. This is consistent with previous findings, as significance of paint in replacement emissions has been indicated by other researchers. (Morales et al., 2020; Tavares, 2006)

Material lifetime multiplier shows a strong relationship with the GWP impact of module B4 (regression analysis in Table 9), strongly decreasing the impact with increasing material lifetime. Such a strong influence of material service life on the replacement impacts has been noted by Morales et al. (2020), who focused on impacts associated with replacement of paint and mortar.

It is worth noting that building lifetime has the opposite effect for recurrent embodied emissions than for the initial ones, meaning that with longer lifetime, replacement is responsible for more emissions. This is also seen as the opposite sign of regression coefficients for building lifetime in modules A1-A4 and B4 (Table 9). Although the total material emissions decrease with increasing lifetime (some emissions are fixed for the whole lifetime), the significance of replacement-related emissions increases.

*The remaining modules (A5, C1, C2-C4)* had negligible GWP impacts (Figure 7). These modules refer to construction activities, demolition activities and end-of-life impacts associated with materials. Module A5 is often reported together with module A4 but its impacts are negligible compared to transport-related impacts of module A4. (Gomes et al., 2018) A study on an office building in Thailand has also shown negligible contribution of construction processes to the total impact. (Kofoworola and Gheewala, 2009) Stage C is also commonly found to be insignificant for the overall emission performance. (Evangelista et al., 2018; Gomes et al., 2018; Morales et al., 2019)

## 5.2 Importance of parameters influencing carbon footprint

The second research question concerned the importance of different parameters on the carbon footprint of Brazilian office buildings. The relationship between carbon footprint and the investigated parameters can be observed from the boxplots shown in Figure 13 and Figure 14 and from the results of a multivariate regression analysis shown in Table 9. The factors are analyzed in the decreasing order of importance according to the regression coefficients for total GWP impacts.

The choice of city had a very strong impact on emission performance of the buildings (Figure 13). However, the first look at Figure 13 shows no apparent trend between these two variables, which can be opposed to a clear trend visible in Figure 5 between energy loads and cities. What is particularly surprising is that Recife, Salvador, Fortaleza have notably higher emissions. An explanation for this finding is that all these three cities are located in the North-eastern region of Brazil, whose electricity is almost twice as emission-intensive than electricity found in the other four regions (Appendix 5). This fact can also

be observed in Figure 8, where highly emission-intensive North-eastern buildings are clearly distinct from the other datapoints.

Therefore, warm climate combined with emission-intensive electricity of cities located in the North-east makes their office building stock particularly polluting. This result is consistent with findings of other researchers, showing that electricity mix is the most important parameter for GWP impacts. (Heeren et al., 2015) Results of the regression analysis clearly show that electricity mix has a much stronger influence on the total GWP impacts than climatic conditions represented using CDH.

Following electricity mix and climate, cooling efficiency has the most influence on the total GWP values, as seen in the regression results. The reason is that module B6 is generally responsible for the highest share of total emissions and cooling efficiency directly influences the delivered energy values considered in this module.

Next, the total GWP is sensitive to WWR and window opening effective area to a very similar degree. The source of this impact can be traced back to modules A1-A4 and B4 in case of WWR and to module B6 in case of window opening effective area.

As we see in Figure 14, WWR-GWP relationship shows a very irregular behavior. The likely reason is the amplification of sampling error seen already for the energy load values in Figure 6, exhibiting similar irregularities. The influence of WWR on the total GWP values is mostly related to high aluminum-related impacts observed in modules A1-A4 and B4.

The influence of window opening effective area on GWP has to be interpreted with caution. It results purely from the fact that this parameter has a strong influence on energy loads, as shown in Table 8. However, this factor is only important for MM buildings. The correlation between the final GWP impacts and window opening effective area is relatively stronger because MM buildings are overrepresented in the sample: on average, there is 1 fully AC building for every 5 MM buildings. However, MM buildings are likely a minority within the office building stock in Brazil.

Further, there is material lifetime multiplier and building lifetime impacts on the GWP value. The regression results show that increase in these parameters should have an increasing effect on the total GWP. However, the trends observed in Figure 14 suggest that there is an increase in emissions for the longest building and material lifetime. As it turns out, this is actually an unfortunate result of sampling error. As can be seen in Figure 6, there seems to be an increasing trend for energy load with increasing lifetime of buildings and components, despite the fact that these parameters were only included <u>after</u> energy modeling. This effect likely influences the total emissions. If the building sample was larger than 1000 items, such sampling errors would decrease and there could be a stronger correlation seen between these lifetime parameters and the total GWP.

Building type, represented by binary variables CL-8 and CL-16, shows a relatively small influence on the total GWP impacts. This parameter has a much stronger influence on the material-related modules of the building life cycle (A1-A4 and B4) but its importance decreases for the total emissions due to the dominance of operational energy use impacts. However, this parameter would gain on importance for a building designed in high standard of energy-efficiency.

Lastly, parameters such as cooling setpoint, SHGC and shading turn out to influence the climate change impacts of the building only to a limited degree. This finding is somehow unexpected as many energy-related studies underlined the importance of these

parameters to energy performance on office buildings, which was the primary reason of including these parameters in the analysis. This shows that parameters important for energy-efficiency of a building do not necessarily have a strong influence the carbon footprint, as there are multiple other parameters that influence the final results. However, the effect of these variables on the GWP impact is still non-negligible.

### 5.3 Recommendations for better emission performance

The third question in this work was related to recommendations that could be given to improve the emission performance of the current office building stock in Brazil.

The study found a very strong correlation between electricity mix and total GWP impacts of the investigated building. North-east region with the most polluting electricity mix happens to also have high energy use for cooling due to warm climate. The exact emission intensity of electricity is different on the state level, with some states worse than others. For example, the state of Ceará, where Fortaleza is located, has its electricity mix of 48% coal, 27% wind, 19% gas, and 6% fuel oil i.e. three quarters of electricity is fossil fuel based. (MME, 2016) As mentioned in the literature review, the electricity production in Brazil is based on hydropower (65% in 2018). (MME, 2019) However, the example of Fortaleza and other cities in the North-east shows that in some regions, renewable energy has a much smaller share, leading to high emission-intensity of buildings. What is also concerning is that the electricity mix in Brazil is steadily deteriorating, suggesting that the whole office building stock is increasingly more polluting with each year. This was already described in the literature review and some possible solutions were also suggested.

The two most important recommendations for improving the GWP performance of the Brazilian office buildings are the same as suggestions for issues with deteriorating quality of electricity mix in Brazil: renewable energy development and efforts to limit the growth of electricity consumption. The best result could be obtained by a combination of the two, because increasing national electricity consumption is one of the main reasons for increasing dependence of the country on fossil fuels for power generation.

As already stated in the background chapter, research efforts have contributed to the creation of national initiatives aimed at increasing the awareness of the most beneficial energy-efficiency strategies, with some national programs focused on office buildings in particular. (CBCS, n.d.; PBE, n.d.; PROCEL, n.d.; ProjetEEE, 2020) In the context of the findings presented in this work, these initiatives have a great importance also for emission performance of the stock, influencing it in at least two ways: 1) a direct influence on emission levels caused by smaller consumption of electricity, responsible for the operational energy use impacts of the buildings; 2) an indirect influence caused by smaller pressure on the electricity grid, which on a larger scale could positively influence the electricity mix.

From this perspective, any parameter that was shown to have a strong correlation with operational energy use could potentially be a part of solution to the problem of increasing emission-intensity of the Brazilian electricity mix. Decreasing cooling setpoint temperature, solar heat gain coefficient of glazing and window-to-wall ratio could all be examples of efforts that could be undertaken in all office buildings, regardless of ventilation mode. Additionally, mixed-mode buildings could be promoted as their emission performance is significantly better, particularly if the windows have a large opening area.

## 5.4 Limitations of the study

The main criticism of this work could be that it does not include national life cycle inventory data for some of the most important processes included in the system, particularly aluminum production. It has been shown by other researchers that data regionalized to Brazilian context can result in significantly smaller environmental impacts than global data. (Morales et al., 2019, 2020) A possible explanation for this trend is a high share of renewable energy in the average Brazilian electricity mix. Therefore, material production impacts can potentially be overestimated.

Adaption of national-specific datasets should also include country-specific transportation data, as many researchers note that Brazilian transportation system is very distinct from the one found in the international database. (Condeixa et al., 2014; Gomes et al., 2018; Taborianski and Prado, 2012) Primarily based on road freight, Brazilian transportation can significantly increase material impacts. (Gomes et al., 2018; Taborianski and Prado, 2012) As a result, higher transportation impacts could partly counterbalance higher material production impacts. Nonetheless, the exact impact of data regionalization in the context of this work can only be found by means of another study.

Another weak point of this work is only a limited consideration of qualities of a mixed-mode ventilation mode. Such systems are strongly dependent on design features such as office partitions (Mariana, 2013) or narrow plan depths, needed for sufficient flow of air across interior spaces (Wood and Salib, 2013, p. 165). Potentially, it could turn out that the considered office geometry does not fit the design requirements for a mixed-mode building. The thermal comfort associated with these two modes was not analyzed either. However, this is partly an intrinsic issue of any study that attempts a comparison between these different ventilation systems, raised for example by Borgstein et al. (2016), who gave Brazilian fully air-conditioned and mixed-mode buildings as an example of these comparability issues.

Likewise, the differences between these two ventilation modes were not fully explored – in real life, each of them can have particular effects on building design and material composition. This would be especially true if a fully naturally ventilated building were considered as well because no AC system would be needed.

Additionally, this work does not consider all the constituents of carbon footprint of a building. Building systems such as drains, ventilation ducts, and fire protection equipment were not included. For material modeling, only elevators and AC units were included, while in real buildings more built-in systems are found. On top of that, some electricity end-uses were not taken into account, e.g. data centers or uninterruptible power sources (UPS), commonly found in Brazil due to regular power fluctuations in the grid. (Borgstein and Lamberts, 2014)

Some more criticism could be directed towards material intensity values which were relatively similar in the considered archetypes, while real buildings are characterized by large variations in material intensity. (De Wolf, n.d.; Heeren and Fishman, 2019)

Finally, the material contribution of shading devices was not considered in this study. Shading devices found in real buildings can be made of aluminum (Brugnera, 2018, p. 149) so their impact on environmental performance on the building could be significant. Incorporating this into the study would also result in interesting trade-off effects related to shading – it would reduce operational energy impacts while at the same time it would increase the embodied material impacts.

## 5.5 Further research

Despite these promising results, some work remains to be done. First of all, the issues mentioned as limitations of this study could be addressed. This would include an implementation of national life cycle inventories, a full analysis of the differences between the ventilation strategies (fully AC vs. MM), and an addition of material composition data (particularly for shading devices).

To develop a full picture of impacts associated with Brazilian office buildings, additional studies should consider more environmental impact categories besides GWP.

Finally, more data collection efforts could allow to estimate parameter distributions found in real buildings, which would yield results representative for the whole office building stock. In such a study, state-level electricity mix could be used to better estimate the GHG emissions associated with electricity use. Such a characterization of the office building stock could serve as a groundwork for possible future studies, allowing for scenario modeling, and assisting in deployment of low-emission pathways. This could also be complemented by an investigation of factors causing poor environmental performance of these buildings, which would propose potential solutions to the most common issues – as it was found out during the course of this work, such poor performance may often result from forces that are far beyond the control of the building designer or facility manager.

# 6 Conclusions

The purpose of the current study was to investigate the carbon footprint of Brazilian office buildings. As it was shown by means of a literature review, this topic has not been yet explored so the climate change impacts associated with the office building stock are not understood. The importance of such a study is supported by numerous environmental issues caused by the current state of the Brazilian construction industry, particularly considering the fact that inaction will make the problems aggravate at an even faster rate due to trends of increasing office floor area and energy use of the sector.

To explore the topic of the carbon footprint of Brazilian office buildings, this work intended to find out what is the range of carbon footprint values of these buildings, which parameters have the strongest impact and how this impact could be reduced.

This was done by means of methodology which included 6 steps: archetype definition, sample selection, energy modeling, material modeling, emission modeling and sensitivity analysis of the results. Archetype definition involved the specification of material composition, basic design features, and operational regimes. The archetypes were created based on the available literature, with some features resembling those typically found in real buildings (such as cooling efficiency), while others are chosen to investigate possible improvement options (such as shading). The archetype definition included 10 variable parameters which were explored using the Latin hypercube sampling method. By means of energy and material modeling, an LCA model was built to assess the climate change impacts associated with the sampled buildings.

The results of the model have shown that the total GWP values vary from 20 to 108 kg  $CO_2$ -eq/m<sup>2</sup>/year, with 41 kg  $CO_2$ -eq/m<sup>2</sup>/year median. The main contributor, and the main source of variations, was the operational use phase (B6), with a median impact of 25  $CO_2$ -eq/m<sup>2</sup>/year. This phase, in turn, was mainly influenced by cooling energy loads, strongly dependent on climatic conditions. The delivered energy was in line with available empirical data. However, the calculated emissions associated with operational energy use could not be benchmarked against any other study.

The life cycle GWP impacts were also influenced by emissions embodied in material. Initial embodied emissions caused by material production and transport to the building site (modules A1-A4) had a median of around 8  $CO_2$ -eq/m<sup>2</sup>/year. This value was found to be of the same order of magnitude as values found in other studies, both in Brazil and abroad. The impact of modules A1-A4 was found to be dominated by emissions associated with aluminum, followed by steel, concrete, lime and brick. The predominant role of aluminum can be supported by findings of other researchers which investigated façade systems typical for office buildings.

Further, the climate change impacts were influenced by recurrent embodied emissions from replacement (module B), whose median impact was around 6  $CO_2$ -eq/m<sup>2</sup>/year. Such order of magnitude of these impacts was also found in another Brazilian study. These material replacement emissions were dominated by aluminum and paint, followed by AC devices, floor covering, doors, and ceramic tile. High significance of paint in replacement emissions is consistent with previous findings.

The GWP impacts associated with construction activities, demolition activities and end-oflife impacts associated with materials (modules A5, C1, C2-C4) were negligible.

The carbon footprint was mainly influenced by the choice of city, which was associated with both with the change of climate (cooling loads) and with emission intensity of electricity. The highest impact values were observed for cities located in the North-east, a region which combines warm climate with a relatively polluting electricity mix. Other parameters with strong influence on the total GWP impacts include cooling efficiency, window opening effective area, and WWR.

Based on these findings, it was suggested that most efforts towards improving the carbon footprint of the Brazilian office buildings should focus on reducing the emission-intensity of the electricity mix, which can be done by increasing the share of renewable energy and by energy-efficiency efforts. In this sense, every parameter that was found to influence operational energy use of the buildings could potentially be a part of the solution. Promotion of mixed-mode buildings was also suggested as a means of decreasing impacts through decreased operational energy use.

The present study provided the first comprehensive assessment of the possible range of carbon footprint values of Brazilian office buildings. It contributes to our understanding of parameters which influence climate change impacts of these buildings during their life cycle stages: construction, operation, and demolition. This new understanding helps to identify strategies that could be used to decrease the environmental impact while still providing the necessary service level needed by the building occupants.

# References

- ABNT, 2013. NBR 15575-1. Edificações Habitacionais Desempenho. Parte 1: Requisitos gerais [Residential buildings Design. Part 1: General requirements]. Associação Brasileira de Normas Técnicas.
- ABNT, 2008. NBR 16401-3. Instalações de ar-condicionado Sistemas centrais e unitários. Parte 3: Qualidade do ar interior [Air conditioning installations – central systems and units. Part 3: Indoor air quality]. Associação Brasileira de Normas Técnicas.
- ABNT, 2005. NBR 12721-2005. Avaliação de custos de construção para incorporação imobiliária e outras disposições para condomínios edilícios [Construction cost assessment for real estate development and other provisions for building condominiums]. Associação Brasileira de Normas Técnicas.
- ABNT, 2003. NBR 15220-2005. Desempenho térmico de edificações [Thermal performance of buildings]. Associação Brasileira de Normas Técnicas.
- Agopyan, V., Souza, U.E.L., Paliari, J.C., Andrade, A.C., 1998. Alternativas para a redução do desperdício de materiais nos canteiros de obras [Alternatives for reducing material waste at construction sites]. Universidade de São Paulo, Rio de Janeiro.
- Alves, T., 2017. Energy savings potential of the high-rise office building stock. A case study of Belo Horizonte, Brazil. (Doctoral thesis). Universidade Federal de Minas Gerais, Belo Horizonte.
- Alves, T., Machado, L., de Souza, R.G., de Wilde, P., 2018. Assessing the energy saving potential of an existing high-rise office building stock. Energy Build. 173, 547– 561. https://doi.org/10.1016/j.enbuild.2018.05.044
- Alves, T., Machado, L., de Souza, R.G., de Wilde, P., 2017. A methodology for estimating office building energy use baselines by means of land use legislation and reference buildings. Energy Build. 143, 100–113. https://doi.org/10.1016/j.enbuild.2017.03.017
- André, M.A. de, 2019. Potencial de incremento do conforto térmico dos usuários em escritórios com o uso de ventiladores de mesa durante o verão [Potential to increase the thermal comfort of users in offices with the use of desk fans during the summer] (Master's thesis). Universidade Federal de Santa Catarina, Florianópolis.
- Andreasi, W.A., 2009. Método para avaliação de conforto térmico em região de clima quente e úmido do Brasil [Method for assessing thermal comfort in hot and humid climate region of Brazil] (Doctoral thesis). Universidade Federal de Santa Catarina, Florianópolis.
- ANVISA, 2003. Diário Oficial da União- Seção 1. Resolução-RE Nº 9 [Official journal of the Union Section 1. Resolution-RE No. 9].
- ASHRAE, 2017. Standard 55-2017: Thermal Environmental Conditions for Human Occupancy. ASHRAE, Atlanta.
- ASHRAE, 2005. 2005 ASHRAE Handbook: Fundamentals. ASHRAE, Atlanta.
- Bendixen, M., Best, J., Hackney, C., Iversen, L.L., 2019. Time is running out for sand. Nature 571, 29–31. https://doi.org/10.1038/d41586-019-02042-4

- Bento, R., Cardoso, P., Komesu, A., Ometto, A., Rossi, E., Rossignolo, J., 2013. Análise do desempenho ambiental de estruturas de concreto armado: uso da Avaliação do Ciclo de Vida (ACV) [Analysis of the environmental performance of concrete structures: use of Life Cycle Assessment (LCA)]. https://doi.org/10.13140/2.1.3142.3040
- Bento, R.C., 2017. Análise do desempenho ambiental de estruturas de concreto armado: uso da avaliação do ciclo de vida (ACV) no processo decisório do dimensionamento (Doutorado em Arquitetura, Urbanismo e Tecnologia). Universidade de São Paulo, São Carlos. https://doi.org/10.11606/T.102.2017.tde-23012017-101051
- Big Ladder Software, n.d. Using Multipliers (Zone and/or Window): Tips and Tricks for Using EnergyPlus — EnergyPlus 8.3 [WWW Document]. URL https://bigladdersoftware.com/epx/docs/8-3/tips-and-tricks-usingenergyplus/using-multipliers-zone-and-or-window.html (accessed 6.4.20).
- BOMA, 2010. Preventive Maintenance Guidebook. Best Practices to Maintain Efficient and Sustainable Buildings, 3rd edition. ed. Building Owners and Managers Association (BOMA) International, Washington D.C.
- Borgstein, E.H., Lamberts, R., 2014. Developing energy consumption benchmarks for buildings: Bank branches in Brazil. Energy Build. 82, 82–91. https://doi.org/10.1016/j.enbuild.2014.07.028
- Borgstein, E.H., Lamberts, R., Hensen, J.L.M., 2016. Evaluating energy performance in non-domestic buildings: A review. Energy Build. 128, 734–755. https://doi.org/10.1016/j.enbuild.2016.07.018
- Brasileiro, L.L., Matos, J.M.E., Brasileiro, L.L., Matos, J.M.E., 2015. Revisão bibliográfica: reutilização de resíduos da construção e demolição na indústria da construção civil [Literature review: reuse of construction and demolition waste in the construction industry]. Cerâmica 61, 178–189. https://doi.org/10.1590/0366-69132015613581860
- Bring, J., 1994. How to Standardize Regression Coefficients. Am. Stat. 48, 209–213. https://doi.org/10.2307/2684719
- Brito, A.C. de, 2015. Contribuição da inércia térmica na eficiência energética de edifícios de escritórios na cidade de São Paulo [Effect of the thermal inertia on the energy performance of office buildings in the city of São Paulo] (Doctoral thesis). Universidade de São Paulo. https://doi.org/10.11606/T.3.2016.tde-13062016-102613
- Brugnera, R.R., 2018. Análise integrada de desempenho energético, impacto ambiental e custo: estudo de soluções de fachada para edifícios de escritórios no Brasil [Integrated analysis of energy performance, environmental impact and cost: study of façades solutions for office buildings in Brazil] (Doctoral thesis).
  Universidade de São Paulo. https://doi.org/10.11606/T.102.2018.tde-31082018-154505
- Brugnera, R.R., 2014. Potencial de economia de energia em edifícios de escritórios com estratégias de ventilação híbrida [Potential of energy savings in office buildings with hybrid ventilation strategies] (Master's thesis). Universidade de São Paulo. https://doi.org/10.11606/D.102.2014.tde-30102014-152311
- Brunner, P.H., Rechberger, H., 2004. Practical Handbook of Material Flow Analysis. Lewis Publishers, Boca Raton, FL.
- Cândido, C.M., 2010. Indoor air movement acceptability and thermal comfort in hothumid climates (Doctoral thesis). Universidade Federal de Santa Catarina, Florianópolis.

- Capistrano, G. de O.W. de, 2008. Eficiência e forma de protetores solares externos em edifícios de escritórios: estudo de casos em Curitiba-PR [Efficiency and form of external solar protector devices for office buildings: a study of cases in Curitiba-PR] (text). Universidade de São Paulo. https://doi.org/10.11606/D.18.2008.tde-05122008-151122
- Carlo, J.C., 2008. Desenvolvimento de metodologia de avaliação da eficiência energética do envoltório de edificações não-residenciais [Development of a methodology for evaluating the energy efficiency of envelopes of non-residential buildings] (Doctoral thesis). Universidade Federal de Santa Catarina, Florianópolis.
- Carvalho, M.M.Q., La Rovere, E.L., Gonçalves, A.C.M., 2010. Analysis of variables that influence electric energy consumption in commercial buildings in Brazil. Renew. Sustain. Energy Rev. 14, 3199–3205. https://doi.org/10.1016/j.rser.2010.07.009
- CBCS, 2016. Benchmarking de Energia [WWW Document]. URL http://benchmarkingenergia.cbcs.org.br/ (accessed 5.13.20).
- CBCS, n.d. Desempenho Energético Operacional em Edificações [WWW Document]. URL http://www.cbcs.org.br/website/benchmarking-energia/show.asp (accessed 5.10.20).
- CEN, 2012. European Standard EN 15978. Sustainability of construction works -Assessment of environmental performance of buildings - Calculation method. European Committee for Standardization.
- Chau, C.K., Hui, W.K., Ng, W.Y., Powell, G., 2012. Assessment of CO2 emissions reduction in high-rise concrete office buildings using different material use options. Resour. Conserv. Recycl. 61, 22–34. https://doi.org/10.1016/j.resconrec.2012.01.001
- Chen, Y., Hong, T., 2018. Impacts of building geometry modeling methods on the simulation results of urban building energy models. Appl. Energy 215, 717–735. https://doi.org/10.1016/j.apenergy.2018.02.073
- Chrisman, L., 2014. Latin Hypercube vs. Monte Carlo Sampling [WWW Document]. Analytica. URL https://lumina.com/latin-hypercube-vs-monte-carlo-sampling/ (accessed 6.1.20).
- CIBSE, 2004. Guide F Energy Efficiency in Buildings. Chartered Institute of Building Services Engineers.
- Condeixa, K., Haddad, A., Boer, D., 2017. Material flow analysis of the residential building stock at the city of Rio de Janeiro. J. Clean. Prod. 149, 1249–1267. https://doi.org/10.1016/j.jclepro.2017.02.080
- Condeixa, K., Haddad, A., Boer, D., 2014. Life Cycle Impact Assessment of masonry system as inner walls: A case study in Brazil. Constr. Build. Mater. 70, 141–147. https://doi.org/10.1016/j.conbuildmat.2014.07.113
- Costa, O., Fuerst, F., Robinson, S.J., Mendes-Da-Silva, W., 2018. Green label signals in an emerging real estate market. A case study of Sao Paulo, Brazil. J. Clean. Prod. 184, 660–670. https://doi.org/10.1016/j.jclepro.2018.02.281
- da Rocha, C.G., Sattler, M.A., 2009. A discussion on the reuse of building components in Brazil: An analysis of major social, economical and legal factors. Resour. Conserv. Recycl. 54, 104–112. https://doi.org/10.1016/j.resconrec.2009.07.004
- de Melo, C.A., Jannuzzi, G. de M., Ferreira Tripodi, A., 2013. Evaluating public policy mechanisms for climate change mitigation in Brazilian buildings sector. Energy Policy 61, 1200–1211. https://doi.org/10.1016/j.enpol.2013.06.056
- de Souza, U.E.L., Paliari, J.C., de Andrade, A.C., Agopyan, V., 2013. Perdas de materiais nos canteiros de obras: a quebra do mito [Material losses at construction sites: breaking the myth].

- De Vecchi, R., 2015. Avaliação de conforto térmico em edificações comerciais que operam sob sistemas mistos de condicionamento ambiental em clima temperado e úmido [Evaluation of thermal comfort in commercial buildings operating under mixed-mode conditioning systems in temperate and humid climates] (Doctoral thesis). Universidade Federal de Santa Catarina, Florianópolis.
- De Wolf, C., 2014. Material quantities in building structures and their environmental impact (Master's thesis). Massachusetts Institute of Technology.
- De Wolf, C., n.d. deQo: database of embodied Quantity outputs [WWW Document]. URL https://www.carbondeqo.com/ (accessed 6.9.20).
- Deetman, S., Marinova, S., van der Voet, E., van Vuuren, D.P., Edelenbosch, O., Heijungs, R., 2020. Modelling global material stocks and flows for residential and service sector buildings towards 2050. J. Clean. Prod. 245, 118658. https://doi.org/10.1016/j.jclepro.2019.118658
- DOE, n.d. EnergyPlus [WWW Document]. URL https://energyplus.net/ (accessed 6.2.20).
- Doka, G., 2009. Life Cycle Inventories of Waste Treatment Services. ecoinvent report No. 13. Swiss Centre for Life Cycle Inventories, Dübendorf.
- Ecoinvent, n.d. Methodology of ecoinvent 3. What is a market and how is it created? [WWW Document]. URL https://www.ecoinvent.org/support/faqs/methodology-ofecoinvent-3/what-is-a-market-and-how-is-it-created.html (accessed 12.16.19).
- Eletrobrás, 2008. Pesquisa de Posse de Equipamentos e Hábitos de Uso. Ano base 2005. Classe Comercial - alta tensão [Survey of Equipment Ownership and Usage Habits. Case year 2005. Commercial Class - high voltage].
- Ellis, P.G., Torcellini, P.A., 2005. Simulating Tall Buildings Using EnergyPlus. Presented at the Ninth International Building Performance Simulation Association (IBPSA) Conference and Exhibition (Building Simulation 2005), Montreal, Quebec, p. 12.
- Evangelista, P.P.A., Kiperstok, A., Torres, E.A., Gonçalves, J.P., 2018. Environmental performance analysis of residential buildings in Brazil using life cycle assessment (LCA). Constr. Build. Mater. 169, 748–761. https://doi.org/10.1016/j.conbuildmat.2018.02.045
- Fialho, R.N., 2007. Edifícios de escritórios na cidade de São Paulo [Office buildings in São Paulo] (Doctoral thesis). Universidade de São Paulo, São Paulo.
- Figueiredo, C.M. de, 2007. Ventilação natural em edifícios de escritórios na cidade de São Paulo: limites e possibilidades do ponto de vista do conforto térmico [Natural ventilation in São Paulo office buildings: thermal comfort limits and possibilities] (Master's thesis). Universidade de São Paulo, São Paulo.
- Foraboschi, P., Mercanzin, M., Trabucco, D., 2014. Sustainable structural design of tall buildings based on embodied energy. Energy Build. 68, 254–269. https://doi.org/10.1016/j.enbuild.2013.09.003
- Forbo, 2016. Carpet Tiles brochure. Forbo Flooring Systems.
- Formoso, C.T., Soibelman, L., De Cesare, C., Isatto, E.L., 2002. Material Waste in Building Industry: Main Causes and Prevention. J. Constr. Eng. Manag. 128, 316– 325. https://doi.org/10.1061/(ASCE)0733-9364(2002)128:4(316)
- Fossati, M., Lamberts, R., Silva, V., 2008. Sustainability in office building design assessment methodology: the case of Florianópolis, SC, Brazil. Presented at the World Conference Sustainable Building 2008, Melbourne, Australia.
- Frischknecht, R., Birgisdottir, H., Chae, C.-U., Lützkendorf, T., Passer, A., Alsema, E., Balouktsi, M., Berg, B., Dowdell, D., Martínez, A.G., Habert, G., Hollberg, A., König, H., Lasvaux, S., Llatas, C., Rasmussen, F.N., Peuportier, B., Ramseier, L., Röck, M., Verdaguer, B.S., Szalay, Z., Bohne, R.A., Bragança, L., Cellura, M.,
Chau, C.K., Dixit, M., Francart, N., Gomes, V., Huang, L., Longo, S., Lupíšek, A., Martel, J., Mateus, R., Ouellet-Plamondon, C., Pomponi, F., Ryklová, P., Trigaux, D., Yang, W., 2019. Comparison of the environmental assessment of an identical office building with national methods. IOP Conf. Ser. Earth Environ. Sci. 323, 012037. https://doi.org/10.1088/1755-1315/323/1/012037

- Gomes, V., Saade, M., Lima, B., Silva, M., 2018. Exploring lifecycle energy and greenhouse gas emissions of a case study with ambitious energy compensation goals in a cooling-dominated climate. Energy Build. 173, 302–314. https://doi.org/10.1016/j.enbuild.2018.04.063
- Häfliger, I.-F., John, V., Passer, A., Lasvaux, S., Hoxha, E., Saade, M.R.M., Habert, G., 2017. Buildings environmental impacts' sensitivity related to LCA modelling choices of construction materials. J. Clean. Prod. 156, 805–816. https://doi.org/10.1016/j.jclepro.2017.04.052
- Heeren, N., 2019. BuildME: Framework to calculate building material & energy expenditures (Source code) [WWW Document]. URL https://github.com/nheeren/BuildME
- Heeren, N., Fishman, T., 2019. A database seed for a community-driven material intensity research platform. Sci. Data 6, 1–10. https://doi.org/10.1038/s41597-019-0021-x
- Heeren, N., Mutel, C.L., Steubing, B., Ostermeyer, Y., Wallbaum, H., Hellweg, S., 2015. Environmental Impact of Buildings—What Matters? Environ. Sci. Technol. 49, 9832–9841. https://doi.org/10.1021/acs.est.5b01735
- Hoxha, E., Habert, G., Chevalier, J., Bazzana, M., Le Roy, R., 2014. Method to analyse the contribution of material's sensitivity in buildings' environmental impact. J. Clean. Prod. 66, 54–64. https://doi.org/10.1016/j.jclepro.2013.10.056
- Hunter, J.D., 2007. Matplotlib: A 2D graphics environment. Comput. Sci. Eng. 9, 90–95. https://doi.org/10.1109/MCSE.2007.55
- Hygh, J.S., DeCarolis, J.F., Hill, D.B., Ranji Ranjithan, S., 2012. Multivariate regression as an energy assessment tool in early building design. Build. Environ. 57, 165– 175. https://doi.org/10.1016/j.buildenv.2012.04.021
- IBGE, 2019a. PNAD Contínua 2018 Características adicionais do mercado de trabalho [PNAD Contínua - 2018 Additional characteristics of the labour market].
- IBGE, 2019b. PNAD Contínua 2018 Características gerais dos domicílios e dos moradores [PNAD Contínua - 2018 General characteristics of households and residents].
- IBGE, 2011. Censo 2010 [Census 2010] [WWW Document]. URL https://censo2010.ibge.gov.br/ (accessed 5.30.20).
- ICCA, 2012. ICCA Building Technology Roadmap. International Council of Chemical Associations.
- IEA, 2019a. Perspectives for the Clean Energy Transition The Critical Role of Buildings. International Energy Agency, Paris.
- IEA, 2019b. Material efficiency in clean energy transitions. International Energy Agency, Paris.
- IEA, 2018. 2018 Global Status Report: Towards a zero-emission, efficient and resilient buildings and construction sector. International Energy Agency and the United Nations Environment Programme.
- IEA, 2017. Energy technology perspectives 2017: Catalysing energy technology transformations. International Energy Agency.

- Inmetro, 2017. Tabelas de consumo/eficiência energética [Consumption/energy efficiency tables] [WWW Document]. URL http://www.inmetro.gov.br/consumidor/pbe/condicionadores.asp (accessed 3.6.20).
- Institut Wohnen und Umwelt GmbH, 2017. Joint EPISCOPE and TABULA Website [WWW Document]. URL http://episcope.eu
- Invidiata, A., Ghisi, E., 2016. Impact of climate change on heating and cooling energy demand in houses in Brazil. Energy Build. 130, 20–32. https://doi.org/10.1016/j.enbuild.2016.07.067
- IPCC, 2015. Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. IPCC, Geneva, Switzerland.
- IPCC, 2014. Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- John, V.M., Pacca, S.A., Angulo, S.C., Punhagui, K.R.G., Oliveira, L.S., Vergara, Y.L.M., Campos, É.F. de, n.d. Projeto Avaliação de Ciclo de Vida Modular de Blocos e Pisos de Concreto [Modular Life Cycle Assessment Project for Concrete Blocks and Floors]. CBCS.
- Kellenberger, D., 2010. Diesel, burned in building machine, GLO. Allocation, cut-off by classification. Ecoinvent database version 3.6.
- Knauf A/S, 2016. Environmental Product Declaration: Knauf Danoline Cleaneo Markant, Contur, Linear and Danopanel (perforated). The Norwegian EPD Foundation.
- Kofoworola, O.F., Gheewala, S.H., 2009. Life cycle energy assessment of a typical office building in Thailand. Energy Build. 41, 1076–1083. https://doi.org/10.1016/j.enbuild.2009.06.002
- Korolija, I., Marjanovic-Halburd, L., Zhang, Y., Hanby, V.I., 2013. UK office buildings archetypal model as methodological approach in development of regression models for predicting building energy consumption from heating and cooling demands. Energy Build. 60, 152–162. https://doi.org/10.1016/j.enbuild.2012.12.032
- Lam, J.C., Hui, S.C.M., 1996. Sensitivity analysis of energy performance of office buildings. Build. Environ. 31, 27–39. https://doi.org/10.1016/0360-1323(95)00031-3
- Lamberts, R., Borgstein, E., Cursino, A., Schinazi, A., De Dominicis, A., 2015. Benchmarking de escritórios corporativos e recomendações para certificação DEO no Brasil [Corporate Office Benchmarking and DEO Certification Recommendations in Brazil]. Conselho Brasileiro de Construção Sustentável (CBCS), São Paulo.
- Lamberts, R., Cândido, C.M., de Dear, R., De Vecchi, R., 2013. Towards a Brazilian Standard on Thermal Comfort. LABEEE.
- Lamberts, R., Ghisi, E., Ramos, G., 2006. Impactos da Adequação Climática Sobre a Eficiência Energética e o Conforto Térmico de Edifícios de Escritórios no Brasil [Impacts of Climate Adaptation on Energy Efficiency and Thermal Comfort of Office Buildings in Brazil]. LABEEE, Florianópolis.
- Lee, A., 2013. pyDOE: the experimental design package for python.
- Lima, K.M. de, 2012. Desempenho térmico e luminoso de brises externos em escritórios em Maceió-AL [Thermal and luminous performance of brise soleil in offices in Maceió] (Master's thesis). Universidade Federal de Alagoas, Maceió.

- Lima, T., 2010. Qualidade ambiental e arquitetônica em edifícios de escritórios: diretrizes para projetos em Brasília [Environmental and architectural quality in office buildings: guidelines for projects in Brasília] (Doctoral thesis). Universidade de Brasília, Brasília.
- Maciel, A., 2002. Projeto bioclimático em Brasília: estudo de caso em edifício de escritórios [Bioclimatic design in Brasilia: case study in an office building] (Master's thesis). Universidade Federal de Santa Catarina, Florianópolis.
- Marcondes, M., Alucci, M., Gonçalves, J., 2012. Façades and office buildings in São Paulo (Brazil): aiming for thermal comfort and natural ventilation, in: Proceedings of 7th Windsor Conference. Windsor, UK, p. 16.
- Marcondes, M.P., 2010. Soluções projetuais de fachadas para edifícios de escritórios com ventilação natural em São Paulo [Facades design solutions for office buildings with natural ventilation in São Paulo] (Doctoral thesis). Universidade de São Paulo. https://doi.org/10.11606/T.16.2010.tde-10112010-102255
- Mariana, F.B., 2013. Análise numérica da influência de configurações de aberturas da fachada e do mobiliário na ventilação natural em ambientes de escritórios [Numerical analysis of the influence of facade openings and furniture configurations on natural ventilation in office environments] (Master's thesis). Universidade de São Paulo. https://doi.org/10.11606/D.3.2013.tde-16052014-151736
- Maslesa, E., Jensen, P.A., Birkved, M., 2018. Indicators for quantifying environmental building performance: A systematic literature review. J. Build. Eng. 19, 552–560. https://doi.org/10.1016/j.jobe.2018.06.006
- Material Economics, 2018. The Circular Economy: A Powerful Force for Climate Mitigation. Material Economics.
- McKay, M.D., Beckman, R.J., Conover, W.J., 1979. A Comparison of Three Methods for Selecting Values of Input Variables in the Analysis of Output From a Computer Code. Technometrics 21, 239–245. https://doi.org/10.1080/00401706.2000.10485979
- McKinsey & Company, 2009. Pathways to a low-carbon economy for Brazil.
- Medeiros, L.M., Durante, L.C., Callejas, I.J.A., Medeiros, L.M., Durante, L.C., Callejas, I.J.A., 2018. Contribuição para a avaliação de ciclo de vida na quantificação de impactos ambientais de sistemas construtivos [Contribution of life cycle assessment to the quantification of the environmental impacts of construction systems]. Ambiente Construído 18, 365–385. https://doi.org/10.1590/s1678-86212018000200259
- Minku, P.M., 2005. Tipologias construtivas de edifícios de escritório na cidade do Florianópolis-SC [Typologies of office buildings in the city of Florianópolis-SC]. Universidade Federal de Santa Catarina, Florianópolis.
- MME, 2019. BEN Séries Históricas Completas 1970-2018 [BEN complete historical series 1970-2018].
- MME, 2016. Matrizes Eletricas Estaduais: Ano Referencia 2015 [Electricity matrices for states: Reference year 2015].
- Morales, M., Moraga, G., Kirchheim, A.P., Passuello, A., 2019. Regionalized inventory data in LCA of public housing: A comparison between two conventional typologies in southern Brazil. J. Clean. Prod. 238, 117869. https://doi.org/10.1016/j.jclepro.2019.117869
- Morales, M.F.D., Reguly, N., Kirchheim, A.P., Passuello, A., 2020. Uncertainties related to the replacement stage in LCA of buildings: A case study of a structural masonry

clay hollow brick wall. J. Clean. Prod. 251, 119649. https://doi.org/10.1016/j.jclepro.2019.119649

- Morishita, C., Sorgato, M.J., Versage, R., Triana, M.A., Marinoski, D.L., Lamberts, R., 2011. Catálogo de propriedades térmicas de paredes e coberturas [Catalogue of thermal properties of walls and roofs]. LABEEE, Florianópolis.
- Müller, D.B., Liu, G., Løvik, A.N., Modaresi, R., Pauliuk, S., Steinhoff, F.S., Brattebø, H., 2013. Carbon Emissions of Infrastructure Development. Environ. Sci. Technol. 47, 11739–11746. https://doi.org/10.1021/es402618m
- Mutel, C.L., de Baan, L., Hellweg, S., 2013. Two-Step Sensitivity Testing of Parametrized and Regionalized Life Cycle Assessments: Methodology and Case Study. Environ. Sci. Technol. 47, 5660–5667. https://doi.org/10.1021/es3050949
- Najjar, M., Figueiredo, K., Palumbo, M., Haddad, A., 2017. Integration of BIM and LCA: Evaluating the environmental impacts of building materials at an early stage of designing a typical office building. J. Build. Eng. 14, 115–126. https://doi.org/10.1016/j.jobe.2017.10.005
- Najjar, M.K., Figueiredo, K., Evangelista, A.C.J., Hammad, A.W.A., Tam, V.W.Y., Haddad, A., 2019. Life cycle assessment methodology integrated with BIM as a decisionmaking tool at early-stages of building design. Int. J. Constr. Manag. 0, 1–15. https://doi.org/10.1080/15623599.2019.1637098
- Neves, L.O., Hopes, A.P., Chung, W.J., Natarajan, S., 2020. "Mind reading" building operation behaviour. Energy Sustain. Dev. 56, 1–18. https://doi.org/10.1016/j.esd.2020.02.003
- Neves, L.O., Marques, T.H.T., 2017. Building Envelope Energy Performance of High-rise Office buildings in Sao Paulo City, Brazil, in: Procedia Environmental Sciences, Sustainable Synergies from Buildings to the Urban Scale. Presented at the International Conference on Sustainable Synergies from Buildings to the Urban Scale, SBE16, pp. 821–829. https://doi.org/10.1016/j.proenv.2017.03.167
- Neves, L.O., Melo, A.P., Rodrigues, L.L., 2019. Energy performance of mixed-mode office buildings: Assessing typical construction design practices. J. Clean. Prod. 234, 451–466. https://doi.org/10.1016/j.jclepro.2019.06.216
- OECD, 2015. OECD Environmental Performance Reviews: Brazil 2015. OECD Publishing, Paris.
- Owen, A., 2019. Monte Carlo theory, methods and examples [WWW Document]. URL https://statweb.stanford.edu/~owen/mc/
- Pasquali, F.E., Beyer, P.O., Hennes, A.M., Ziebell, C.S., 2011. Potential for energy savings in retrofitting of an office buildings in São Paulo/Brazil, in: Proceedings of Building Simulation. Presented at the 12th Conference of International Building Performance Simulation Association, Sydney.
- Paulsen, J.S., Sposto, R.M., 2013. A life cycle energy analysis of social housing in Brazil: Case study for the program "MY HOUSE MY LIFE." Energy Build. 57, 95–102. https://doi.org/10.1016/j.enbuild.2012.11.014
- PBE, n.d. PBE Programa Brasileiro de Etiquetagem [WWW Document]. URL http://www.pbeedifica.com.br/sobre (accessed 5.10.20).
- Pereira, F.A., 2019. Influência da estratégia de ventilação natural no desempenho termoenergético de edifícios de escritórios de modo misto [Influence of the natural ventilation strategy on the thermoenergetic performance of mixed-mode office buildings] (Master's thesis). Universidade Estadual de Campinas, Campinas.
- PINI, 2010. TCPO: Tabelas de Composições de Preços Para Orçamentos [TCPO: Price Composition Tables for Budgets], 13th ed.

- Primas, A., 2010. Blower and heat exchange unit production, central, 600-1200 m3/h, RoW. Allocation, cut-off by classification. Ecoinvent database version 3.6.
- PROCEL, n.d. Programa Nacional de Conservação de Energia Elétrica [WWW Document]. URL http://www.procelinfo.com.br/main.asp (accessed 5.10.20).
- ProjetEEE, 2020. Projetando Edificações Energeticamente Eficientes [WWW Document]. URL http://projeteee.mma.gov.br/ (accessed 3.6.20).
- Ramesh, T., Prakash, R., Shukla, K.K., 2010. Life cycle energy analysis of buildings: An overview. Energy Build. 42, 1592–1600. https://doi.org/10.1016/j.enbuild.2010.05.007
- Ramos, J., 2010. Efeito da inércia térmica em edifício de escritórios: estudo de caso em Belo Horizonte [Effect of thermal inertia in office building: case study in Belo Horizonte] (Master's thesis). Universidade Federal de Minas Gerais, Belo Horizonte.
- Reis, D. da C., Mack-Vergara, Y., John, V.M., 2019. Material flow analysis and material use efficiency of Brazil's mortar and concrete supply chain. J. Ind. Ecol. 23, 1396– 1409. https://doi.org/10.1111/jiec.12929
- Rohden, A.B., Garcez, M.R., 2018. Increasing the sustainability potential of a reinforced concrete building through design strategies: Case study. Case Stud. Constr. Mater. 9, e00174. https://doi.org/10.1016/j.cscm.2018.e00174
- Roriz, M., 2012. Arquivos Climáticos de Municípios Brasileiros [Climate files of Brazilian municipalities]. Associação Nacional de Tecnologia do Ambiente Construído, São Carlos, Brazil.
- Rupp, R.F., Ghisi, E., 2017. Predicting thermal comfort in office buildings in a Brazilian temperate and humid climate. Energy Build. 144, 152–166. https://doi.org/10.1016/j.enbuild.2017.03.039
- Saade, M., Silva, M.G. da, Gomes, V., Franco, H.G., Schwamback, D., Lavor, B., 2014. Material eco-efficiency indicators for Brazilian buildings. Smart Sustain. Built Environ. https://doi.org/10.1108/SASBE-04-2013-0024
- Santa Rosa Rocha, M., 2019a. Building construction, budget hotel, BR, Allocation at the point of substitution, Ecoinvent database version 3.6.
- Santa Rosa Rocha, M., 2019b. Building construction, luxury hotel, BR, Allocation at the point of substitution, Ecoinvent database version 3.6.
- Santa Rosa Rocha, M., 2019c. Building construction, hostel, BR, Allocation at the point of substitution, Ecoinvent database version 3.6.
- Santana, M., 2006. Influência de parâmetros construtivos no consumo de energia de edifícios de escritório localizados em Florianópolis-SC [Influence of construction parameters on energy consumption of office buildings located in Florianópolis-SC] (Master's thesis). Universidade Federal de Santa Catarina, Florianópolis.
- Santesso, C.A., 2018. parIDF: Python open source code developed to perform parametric simulations through EnergyPlus Building Energy Simulation (BES) software (Source code) [WWW Document]. Universidade São Paulo. URL https://www.iau.usp.br/laboratorios/lca/index.php/trabalhos-conforto/ (accessed 6.2.20).
- Santesso, C.A., Chvatal, K.M.S., 2018. Consumo energético de salas de escritório climatizadas de modo misto em São Paulo: considerações sobre o projeto. Ambiente Construído 18, 119–137. https://doi.org/10.1590/s1678-86212018000400297

- Seto, K.C., Davis, S.J., Mitchell, R.B., Stokes, E.C., Unruh, G., Ürge-Vorsatz, D., 2016. Carbon Lock-In: Types, Causes, and Policy Implications. Annu. Rev. Environ. Resour. 41, 425–452. https://doi.org/10.1146/annurev-environ-110615-085934
- Silva, B.V. da, 2013. Construção de ferramenta para avaliação do ciclo de vida de edificações [Building Tool for Life Cycle Assessment of Buildings] (text). Universidade de São Paulo. https://doi.org/10.11606/D.86.2013.tde-15052013-103100
- Silva, D.A.L., 2012. Avaliação do ciclo de vida da produção do painel de madeira MDP no Brasil [Life cycle assessment of MDP wood-based panel production in Brazil] (Master's thesis). Universidade de São Paulo. https://doi.org/10.11606/D.88.2012.tde-31072012-121351
- Silva, F., Cleto, F., Diestelkamp, E., Yoshida, O., Oliveira, L., Saade, M., Silva, V., Moraga, G., Passuello, A., Silva, M., Myers, N., Gmünder, S., 2018. Life Cycle Inventories of Cement, Concrete and Related Industries - Brazil.
- Sinduscon-MG, 2007. Custo unitário básico (CUB/m<sup>2</sup>): principais aspectos [Basic unit cost (CUB/m<sup>2</sup>): main aspects]. Sinduscon-MG, Belo Horizonte.
- Soares, S.R., Pereira, S.W., 2004. Inventário da produção de pisos e tijolos cerâmicos no contexto da análise do ciclo de vida [Inventory of the production of ceramic tiles and bricks in the context of life cycle analysis]. Ambiente Construído 4, 83–94.
- Sonter, L.J., Herrera, D., Barrett, D.J., Galford, G.L., Moran, C.J., Soares-Filho, B.S., 2017. Mining drives extensive deforestation in the Brazilian Amazon. Nat. Commun. 8, 1–7. https://doi.org/10.1038/s41467-017-00557-w
- Souza, D.M. de, Lafontaine, M., Charron-Doucet, F., Bengoa, X., Chappert, B., Duarte, F., Lima, L., 2015. Comparative Life Cycle Assessment of ceramic versus concrete roof tiles in the Brazilian context. J. Clean. Prod. 89, 165–173. https://doi.org/10.1016/j.jclepro.2014.11.029
- Symeonidis, A., 2019. Market for waste glass, BR. Allocation, cut-off by classification. Ecoinvent database version 3.6.
- Taborianski, V.M., Prado, R.T.A., 2012. Methodology of CO2 emission evaluation in the life cycle of office building façades. Environ. Impact Assess. Rev. 33, 41–47. https://doi.org/10.1016/j.eiar.2011.10.004
- Tamanini Junior, T., Ghisi, E., 2015. Características construtivas de edifícios de escritório localizados em Florianópolis [Construction features of office buildings located in Florianópolis]. Presented at the XIII Encontro Nacional e IX Encontro Latinoamericano de Conforto no Ambiente Construído.
- Tavares, S.F., 2006. Metodologia de análise do ciclo de vida energético de edificações residenciais brasileiras [Methodology for analyzing the energy life cycle of Brazilian residential buildings] (Doctoral thesis). Universidade Federal de Santa Catarina, Florianópolis.
- The Carbon Leadership Forum, 2018. Life Cycle Assessment of Buildings: Technical Guidance. The Carbon Leadership Forum.
- The World Bank, 2019. Population estimates and projections [WWW Document]. DataBank. URL https://databank.worldbank.org/source/population-estimates-andprojections# (accessed 4.28.20).
- Thomaz, E., Filho, C.V.M., Cleto, F. da R., Cardoso, F.F., 2009. Código de práticas No. 01. Alvenaria de vedação em blocos cerâmicos [Code of practice No. 01. Ceramic brick masonry]. Instituto de Pesquisas Tecnológicas do Estado de São Paulo, São Paulo.

- Treloar, G.J., Fay, R., Ilozor, B., Love, P.E.D., 2001. An analysis of the embodied energy of office buildings by height. Facilities 19, 204–214. https://doi.org/10.1108/02632770110387797
- U.S. EIA, 2015. A Look at the U.S. Commercial Building Stock: Results from EIA's 2012 Commercial Buildings Energy Consumption Survey (CBECS) [WWW Document]. URL https://www.eia.gov/consumption/commercial/reports/2012/buildstock/?src=%E2 %80%B9%20Consumption%20%20Commercial%20Buildings%20Energy%2

0Consumption%20Survey%20(CBECS)-b4 (accessed 3.28.20).

- Veloso, A.C. de O., 2017. Avaliação do consumo de energia elétrica de edificações de escritório e sua correlação com as decisões de projeto [Evaluation of electrical energy consumption in office buildings and their correlation with design decisions] (Doctoral thesis). Universidade Federal de Minas Gerais, Belo Horizonte.
- Veloso, A.C. de O., de Souza, R.V.G., Koury, R.N.N., 2017. Research of Design Features that Influence Energy Consumption in Office Buildings in Belo Horizonte, Brazil, in: Energy Procedia. Presented at the 8th International Conference on Sustainability in Energy and Buildings, SEB-16, 11-13 September 2016, Turin, Italy, pp. 101– 110. https://doi.org/10.1016/j.egypro.2017.03.012
- Versage, R., Borgstein, E., Lamberts, R., n.d. Grau-horas de resfriamento GHR [Cooling degree-hours CDH]. Conselho Brasileiro de Construção Sustentável (CBCS).
- Waskom, M., Botvinnik, O., Joel Ostblom, Gelbart, M., Lukauskas, S., Hobson, P., Gemperline, D.C., Augspurger, T., Halchenko, Y., Cole, J.B., Warmenhoven, J., Julian de Ruiter, Cameron Pye, Stephan Hoyer, Vanderplas, J., Santi Villalba, Kunter, G., Quintero, E., Bachant, P., Martin, M., Meyer, K., Swain, C., Miles, A., Brunner, T., O'Kane, D., Tal Yarkoni, Mike Lee Williams, Constantine Evans, Clark Fitzgerald, 2020. Seaborn: Statistical data visualization using matplotlib (source code) [WWW Document]. https://doi.org/10.5281/zenodo.3767070
- Wong, I.L., Krüger, E., Loper, A.C.M., Mori, F.K., 2019. Classification and energy analysis of bank building stock: A case study in Curitiba, Brazil. J. Build. Eng. 23, 259– 269. https://doi.org/10.1016/j.jobe.2019.02.003
- Wood, A., Salib, R., 2013. Natural Ventilation in High-Rise Office Buildings: An output of the CTBUH Sustainability Working Group. Routledge, Abingdon.
- Yoshino, H., Hong, T., Nord, N., 2017. IEA EBC annex 53: Total energy use in buildings— Analysis and evaluation methods. Energy Build. 152, 124–136. https://doi.org/10.1016/j.enbuild.2017.07.038

## Appendices

# Appendix 1: Material intensity for materials and equipment in three building archetypes.

Material/equipment	Material intensity (kg/m2)			Source	Assumed
	OPL-8	CL-8	CL-16	000100	density/weight
Concrete	895.40	885.05	1 247.02	(1)	2380 kg/m <sup>3</sup>
Brick	72.38	102.34	135.94	(1)	2.2 kg/unit
Concrete block	28.37	16.11	13.87	(1)	16.2 kg/unit
Steel	33.31	38.89	58.70	(1)	-
Cement	19.86	16.28	21.51	(1)(2)	-
Lime	49.66	40.70	53.78	(1)(2)	-
Gravel	62.09	45.72	57.29	(1)	1430 kg/m <sup>3</sup>
Sand	324.26	274.44	364.62	(1)	1500 kg/m <sup>3</sup>
Fiber cement roof tile	1.68	2.12	1.30	(1)	1900 kg/m <sup>3</sup>
Plywood	16.13	16.66	20.71	(1)	625 kg/m <sup>3</sup>
Doors	2.10	3.16	4.12	(1)	27.6 kg/m <sup>2</sup>
Ceramic tile	11.20	11.85	16.03	(1)	15.56 kg/m <sup>2</sup>
Glass	according to the window-to-wall ratio			-	15 kg/m <sup>2</sup>
Aluminum	according to the window-to-wall ratio			(3)	50.7 kg/m <sup>2</sup>
Stone	0.65	0.38	0.47	(1)	78 kg/m <sup>2</sup>
Gypsum plasterboard	1.20	1.24	1.81	(1)	6.6 kg/m <sup>2</sup>
Paint	2.31	2.13	3.26	(1)	1.1 kg/liter
Copper	0.49	0.29	0.77	(1)	-
Sanitary ceramics	0.70	0.72	0.72	(1)	25 g/unit
Plastics	0.09	0.10	0.09	(1)	0.24 kg/m
Bitumen emulsion	4.84	2.31	2.88	(1)	-
Floor covering	1.80	1.80	1.80	(4)	-
Acoustic ceiling	11.10	11.10	11.10	(5)	-
Elevator	1.04	1.04	0.78	(6)	2490 kg/unit
AC unit	2.56	2.56	2.56	(7)	143 kg/unit

(1) ABNT (2005)

(2) Cement and lime reported as one value in ABNT (2013); they were separated using a proportion of 1:2.5, assuming that half of the final product is plaster (1:2) and half is mortar (1:3). (Thomaz et al., 2009, p. 41)
(3) Assumed 1:4 proportion of glass to aluminum. (ABNT, 2005)

(4) Forbo (2016)

(5) Knauf A/S (2016)

(6) 8-floor and 16-floor archetypes have 2 and 3 elevators, respectively. (Sinduscon-MG, 2007)

(8) Assumed AC capacity 900 m<sup>3</sup>/h which is average capacity from the Ecoinvent process data (Primas, 2010), airflow 7.5 liter/s/person (see 3.1.4) and occupant density 0.11 people/m<sup>2</sup> (see 3.1.5).

Material/equipment	Element type	Lifetime range (years)	Adopted lifetime (years)	Source
Concrete	Main structure	-	building lifetime	(1)(2)
Brick	Façade	40-building lifetime	building lifetime	(1)(2)
Concrete block	Façade	40-building lifetime	building lifetime	(1)(2)
Steel	Main structure	-	building lifetime	(1)(2)
Cement	Façade	40-building lifetime	building lifetime	(3)
Lime	Façade	40-building lifetime	building lifetime	(3)
Gravel	Other	40-building lifetime	building lifetime	(3)
Sand	Façade	40-building lifetime	building lifetime	(3)
Fiber cement roof tile	Roof	10-100	35	(1)(2)
Plywood	Main structure*	-	building lifetime	(1)(2)
Doors	Windows & Doors	5-20	12	(1)(2)
Ceramic tile	Finishes	13-50	20	(1)(2)
Glass	Windows & Doors	20-60	45	(1)(2)
Aluminum	Windows & Doors	20-60	45	(1)(2)
Stone	Finishes	-	30	(4)
Gypsum plasterboard	Finishes	13-30	20	(1)(2)
Paint	Finishes	3-12	8	(1)(2)
Copper	Other	20-40	30	(1)(2)
Sanitary ceramics	Other	30	30	(2)
Plastics	Other	20-40	30	(1)(2)
Bitumen emulsion	Roof	10-100	35	(1)(2)
Floor covering (carpet)	Finishes	5-12	8	(1)(2)
Acoustic ceiling	Finishes	60	60	(5)
Elevator	Systems	10-50	25	(2)
AC unit	Systems	10-20	15	(2)

### Appendix 2: Lifetimes of materials and equipment used in buildings.

\* Plywood is used as formwork for concrete casting.

(1) ABNT (2013)

(2) BOMA (2010)

(3) Assumed to be a part of the façade.

(4) Marble countertops, assumed 30 years of lifetime

(5) Knauf A/S (2016)

Material/equipment	Production process	Disposal process
Concrete	market for concrete, 30MPa, BR	treatment of inert waste, inert material landfill, RoW
Brick	market for clay brick, GLO	treatment of inert waste, inert material landfill, RoW
Concrete block	market for concrete block, BR	treatment of inert waste, inert material landfill, RoW
Steel	market for reinforcing steel, GLO	treatment of scrap steel, inert material landfill, RoW
Cement	market for cement, unspecified, BR	treatment of inert waste, inert material landfill, RoW
Lime	market for lime, hydrated, packed, RoW	treatment of inert waste, inert material landfill, RoW
Gravel	market for gravel, crushed, BR	treatment of inert waste, inert material landfill, RoW
Sand	market for sand, BR	treatment of inert waste, inert material landfill, RoW
Fiber cement roof tile	market for fibre cement roof slate, GLO	treatment of inert waste, inert material landfill, RoW
Plywood	market for plywood, for indoor use, RoW	market for waste wood, untreated, BR
Doors	market for door, inner, wood, GLO	market for waste wood, untreated, BR
Ceramic tile	market for ceramic tile, GLO	treatment of inert waste, inert material landfill, RoW
Glass	market for flat glass, coated, RoW	market for waste glass, BR
Aluminum	market for window frame, aluminium, U=1.6 W/m2K, GLO	treatment of aluminium scrap, post-consumer, by collecting, sorting, cleaning, pressing, RoW
Stone	market for natural stone plate, cut, GLO	treatment of inert waste, inert material landfill, RoW
Gypsum plasterboard	market for gypsum plasterboard, GLO	treatment of waste gypsum, inert material landfill, RoW
Paint	market for alkyd paint, white, without water, in 60% solution state, RoW	treatment of waste paint, inert material landfill, RoW
Copper	market for wire drawing, copper, GLO	market for copper scrap, sorted, pressed, GLO
Sanitary ceramics	market for sanitary ceramics, GLO	treatment of inert waste, inert material landfill, RoW
Plastics	market for extrusion, plastic pipes, GLO	market for waste plastic, mixture, BR
Bitumen emulsion	market for bitumen adhesive compound, cold, GLO	treatment of waste bitumen, sanitary landfill, RoW
Floor covering	market for textile, non woven polypropylene, GLO	market for waste polypropylene, BR
Acoustic ceiling	no Ecoinvent process, an Environme (Knauf A/S, 2016)	ntal Product Declaration (EPD) used instead
Elevator	market for elevator, hydraulic, GLO	(not applicable – disposal included in the production process)
AC unit	market for blower and heat exchange unit, central, 600-1200 m3/h, GLO	treatment of used blower and heat exchange unit, central, 600-1200 m3/h, RoW

### Appendix 3: LCI datasets for materials and equipment.

\* Ecoinvent 3.6, cut-off allocation. ReCiPe midpoint H method.

Reference name	Process*
Diesel	diesel, burned in building machine, GLO
Machinery	building machine production, RoW
Transport	market for transport, freight, lorry, unspecified, RoW

#### Appendix 4: LCI datasets for construction and end-of-life processes.

\* Ecoinvent 3.6, cut-off allocation. ReCiPe midpoint H method.

# Appendix 5: Selected Brazilian cities, their cooling degree hours (CDH), region, LCI datasets, and elementary GWP impact of the regional electricity.

City	CDH*	Region	Electricity process **	GWP (kg CO2-eq/kWh)***
São Paulo	14172	South-east	market for electricity, low voltage, BR-South-eastern grid	0.227
Rio de Janeiro	45016	South-east	market for electricity, low voltage, BR-South-eastern grid	0.227
Brasília	16624	Mid-west	market for electricity, low voltage, BR-Mid-western grid	0.145
Salvador	67930	North-east	market for electricity, low voltage, BR-North-eastern grid	0.398
Fortaleza	71394	North-east	market for electricity, low voltage, BR-North-eastern grid	0.398
Belo Horizonte	23883	South-east	market for electricity, low voltage, BR-South-eastern grid	0.227
Manaus	82005	North	market for electricity, low voltage, BR-Northern grid	0.197
Curitiba	9397	South	market for electricity, low voltage, BR-Southern grid	0.137
Recife	63550	North-east	market for electricity, low voltage, BR-North-eastern grid	0.398
Goiânia	31081	Mid-west	market for electricity, low voltage, BR-Mid-western grid	0.145
Belém	81393	North	market for electricity, low voltage, BR-Northern grid	0.197
Porto Alegre	23954	South	market for electricity, low voltage, BR-Southern grid	0.137

\* Calculated by Versage et al. (n.d.) with a wet-bulb temperature of 15°C

\*\* Ecoinvent 3.6, cut-off allocation.

\*\*\* Ecoinvent 3.6, cut-off allocation. ReCiPe midpoint H method.



